#### UNIVERSITY OF CALIFORNIA

Los Angeles

## Probing Techniques for Multiuser Channels with Power Control

A dissertation submitted in partial satisfaction of the

requirements for the degree Doctor of Philosophy

in Electrical Engineering

by

**Christopher James Hansen** 

1997

© Copyright by Christopher James Hansen, 1997

All Rights Reserved

The dissertation of Christopher James Hansen is approved.

Eitan Tadmore

William Mangione-Smith

Henry Samueli

Gregory J. Pottie, Committee Chair

University of California, Los Angeles

1997

# **Dedication**

To the memory of my father, David Hansen.

# **Table of Contents**

Dedication	1	iii
Table of C	Contents	iv
List of Fig	ures	vii
List of Tab	bles i	x
Acknowled	dgements	x
VITA		xii
ABSTRAC	CT OF THE DISSERTATION	xiii
1. Introd	luction	1
1.1 Mot	tivation for Channel Probing	1
1.2 Ove	erview of Dissertation Topics	3
2. Backg	ground	5
2.1 Wir	eless Channels	5
2.1.1	Propagation Model	6
2.1.2	Cellular Systems	8
2.2 Wir	eline Channels	
2.2.1	Twisted Pair Cable Characteristics	
2.2.2	Wireline Signaling Format	12
2.2.3	Wireline Channel Access	13
2.3 Mul	ltiuser Channels	14
2.4 Mul	ltiuser Channel Capacity	16
2.4.1	Practical Restrictions on Signaling Format	17

2.4.2	Practical Capacity of Interference Channels	18
2.5 Dyn	amic Channel Allocation	19
2.6 Pow	er Control	20
2.6.1	Fundamental Concepts	20
2.6.2	Distributed Power Control	21
2.7 Sum	imary	22
3. Probin	g Power Controlled Channels	25
3.1 Prob	oing for Maximum Feasible SIR	25
3.2 Prot	bing Accuracy	27
3.2.1	Estimation With Coherent Detection	
3.2.2	Estimation Without Detection	31
3.2.3	Effect of Distributed Power Control	34
3.2.4	Effect of User Dynamics	
3.3 App	lication of Probing to Indoor Wireless Systems	40
3.3.1	Simulation of System with Probing	40
3.3.2	Simulation Results	42
3.4 App	lication of Probing to ADSL Wireline Systems	43
3.4.1	Distributed Access for Wireline Systems	44
3.4.2	ADSL Simulation	45
3.4.3	Simulation Results	48
3.5 Sun	ımary	50
4. Probin	g Time Varying Channels	52
4.1 Cha	nnel Partitioning for Mobile Radio	54
4.1.1	Channel Partitioning Concept	54
4.1.2	Partitioned System Model	60
4.1.3	Trade-offs in System Design	62
4.2 Mol	oility and Hand-off	65
4.3 Cha	nnel Partitioning with Power Level Constraints	68
4.4 Sum	imary	73

5. Probing Channels with Adaptive Antenna Arrays	75
5.1 Adaptive Antenna Systems for Wireless Communication	76
5.2 Adaptive Antennas with Power Control	81
5.3 Transmitter Antenna Weights	83
5.4 Probing with Adaptive Antennas	84
5.4.1 MSE Bound for a New User	85
5.4.2 Probing with Receiver Measurements	87
5.5 Practical Considerations	90
5.6 Simulation Results	92
5.7 Summary	96
6. Conclusions	98
A. Appendix A - MMSE Receiver with Distributed Power Control	101
A.1 Background	101
A.2 Proof	102
A.2.1 Positivity Property (equation (A.4))	103
A.2.2 Monotonicity Property (equation (A.5))	103
A.2.2.1 Case I	103
A.2.2.2 Case II	103
A.2.3 Scalability Property (equation (A.6))	104
B. Appendix B - Glossary	106
B.1 List of Acronyms and Abbreviations	106
B.2 Definitions	108

# **List of Figures**

Figure 2-1	Cellular Concept with Re-use Factor 7	9
Figure 2-2	Loop Gain with Typical NEXT and FEXT Response	13
Figure 2-3	Wireline Multiuser Channel	14
Figure 2-4	Wireless Multiuser Channel	14
Figure 2-5	Interference Channel Model	15
Figure 3-1	Interference Power Estimation	29
Figure 3-2	Probing Error Due to Interference Estimation	34
Figure 3-3	Probing Error From Power Control Convergence	37
Figure 3-4	Channel Probing Performance	39
Figure 3-5	Performance of Indoor Wireless Systems	43
Figure 3-6	ADSL Simulation Results	49
Figure 4-1	Signal Frame Format	55
Figure 4-2	Network Model	61
Figure 4-3	User Capacity Optimization	64
Figure 4-4	Blocking Probability with Mobility	68
Figure 4-5	IS-54 Comparison	72

Figure 5-1	Peer-to-Peer System with Adaptive Antennas	77
Figure 5-2	Transmitter Array	78
Figure 5-3	Receiver Array with Weight Update	78
Figure 5-4	Probing Receiver Structure	88
Figure 5-5	Simulation of Probing Algorithm	89
Figure 5-6	User Location Grid	93

# **List of Tables**

Frequency Hopped System with Probing	41
ADSL Rate Table	44
Simulation Parameters	46
ADSL Revenue Summary	48
TDMA Simulation Parameters	71
Typical Simulation Results	93
Probing Simulation Results, p <sub>target</sub> = 30 dB	95
Probing Simulation Results, p <sub>target</sub> = 35 dB	95

## Acknowledgements

An enormous number of people have helped me and made contributions to this work. I would like to begin by thanking my committee members, especially my advisor, Gregory Pottie, who gave me the idea to investigate channel probing for wireless communications. His insight is greatly appreciated.

Next, I'd like to thank my fellow UCLA students for their support. My research tied in closely with work by Shou Chen, Victor Lin, Charles Wang, and Eldad Perahia and I am very glad for the opportunity to work with them.

I extend great thanks to my fiancee, Grace Siu, for her patience and support. I also wish to thank my mother Ann Hansen, my brother Mark, and my sister, Leigh.

My experience in industry taught me a great deal about engineering and communications and I would like to thank the people I worked with at Bell Labs and Lincoln Labs. Special thanks go out to Nitin Shah, John Rucki, Willy Gilson, Gordon Armstrong, and Bing Chang for their mentoring and encouragement.

From my masters degree work at the University of Massachusetts I would like to acknowledge the help of Christopher Ruf and Jim Mead.

My interest in radio communications began at Rensselaer and more

Х

specifically at WRPI. I'd like to thank the people there who made it such an interesting place to discuss radio and just about anything else. I owe special thanks to Fred Gotwald, Aaron Heller, John Lattyak, Jim Mastracco, Gary Cutbill and Ken Johnson. I also learned a fair bit about radio at W2SZ, the RPI Amateur Radio Club, and I'd like to thank the people I knew there, especially Phillip Langlois.

Last, I'd like to acknowledge the inspiration and support that my father, David Hansen, gave to me. In his memory I have dedicated this dissertation, because I cannot begin to express my gratitude in these words I write today and I wish so much that I could thank him.

### VITA

March 22, 1967	Born, Troy, New York
1987	B.S., Electrical Engineering Rensselaer Polytechnic Institute Troy, New York
1989	M.S., Electrical and Computer Engineering University of Massachusetts Amherst, Massachusetts
1989-1990	Member of Technical Staff Lincoln Laboratory, Massachusetts Inst. of Tech. Lexington, Massachusetts
1991-1992	Member of Technical Staff AT&T Bell Laboratories Whippany, New Jersey

### PUBLICATIONS AND PRESENTATIONS

- C. J. Hansen, C. C. Wang, and G. J. Pottie, "Distributed dynamic channel resource allocation in wireless systems," Asilomar Conference on Signals, Systems, and Computers, Oct. 1994.
- C. J. Hansen and G. J. Pottie, "Distributed Access Control in Wireless and Wireline Systems" IEEE International Symposium on Information Theory, September 1995.

#### **ABSTRACT OF THE DISSERTATION**

## Probing Techniques for Multiuser Channels with Power Control

by

Christopher J. Hansen

Doctor of Philosophy in Electrical Engineering University of California, Los Angeles, 1997 Professor Gregory J. Pottie, Chair

This work investigates a set of probing techniques for multiuser radio systems. These techniques are designed to allow radio transmitter/receiver pairs to determine the characteristics of a communication channel they wish to use. It is assumed that the channel is shared by multiple users that employ power control to allow each pair to maintain a fixed signal-to-interference ratio and transmission rate. The goal is to allow a new user pair to determine the maximum feasible bit rate it can achieve and the information necessary to choose a signaling format that will yield the maximum rate.

Three particular channel models are examined: a high symbol rate indoor channel, a moderate symbol rate mobile radio channel, and a channel with adaptive antenna arrays in addition to power control. The probing techniques are tailored to give information that is appropriate for a system operating under each channel model. The indoor channel probing algorithm gives precise results for the maximum feasible SIR level. The mobile radio probing algorithm quickly determines whether or not a particular channel can be used for communication at a fixed rate. Probing with adaptive antennas gives an estimate of the SIR that can be achieved at a target transmitter power.

The intended applications for channel probing are future wireless systems that are fully distributed. Until now, most distributed access algorithms have used ad hoc techniques. Channel probing puts distributed access on a solid foundation and allows distributed systems to employ sophisticated power and channel allocation techniques that have been developed for centralized systems.

## Chapter 1

## Introduction

This dissertation introduces new techniques for determining the characteristics of communication channels that are shared by multiple users. These techniques are grouped together under the name *channel probing*. As the name suggests, probing takes the form of a single user transmitting a signal on a channel and measuring a response, usually a measurement of both the received signal power and received interference power from other users. From the response, the user is able to determine the characteristics of the channel. This information is in turn used to determine the optimal modulation format and power level for communication.

### **1.1 Motivation for Channel Probing**

Since the introduction of cellular telephone systems, researchers have been investigating methods for increasing the capacity, or number of users per unit of bandwidth, that the systems can handle. This research is critical, because radio bandwidth is limited and the demand for wireless communication services is increasing. At the same time, new wireless communication services, such as two-way pagers, personal communication systems, navigation systems and many others are also being developed and competing for increasingly scarce radio spectrum. This trend is likely to continue for the foreseeable future.

The application of digital technology has brought new cellular standards, IS-54 TDMA (time division multiple access) and IS-95 CDMA (code division multiple access), with large capacity gains relative to present analog systems. However, the capacity increase has exacted its own cost: an enormous increase in complexity. This has already lead to problems in implementing new systems and in convincing the telecommunications community that they are viable solutions. For example, there are no IS-95 CDMA systems in operation in the United States as of November 1996, even though VLSI chip sets to implement the phones have been available since 1991. There are also very few operating TDMA systems.

The problem with these new systems is not the difficulty in implementing the radio modem. In fact, the sophisticated VLSI circuits work extremely well. Rather, the difficulty has been the implementation of complex control networks necessary to make the new systems work. Unless changes are made, future systems - which will be even more complicated - will also encounter complexity problems.

A potential solution to the complexity dilemma is to make complex multiuser

communication systems more distributed. With distributed systems, the choice of channel, signalling format, and power can be made by each user based on its own measurements of the multiuser channel. Recent developments [3][4][18] have demonstrated that power control can be made completely distributed with the same results as a centralized system. Distributed channel allocation, however, is a newer topic, and previous attempts have used a variety of ad-hoc techniques [13][66].

The purpose of probing is to allow each user to measure its own channel characteristics so it can apply the same type of power and channel allocation algorithm that a centralized controller would employ. With probing, users work together to find appropriate channel allocations that help to maximize capacity. Probing helps to formalize a method for dynamic channel allocation. Instead of relying on ad-hoc measures of channel congestion, the channel probe allows the user to precisely determine the maximum communication rate it can achieve.

## **1.2** Overview of Dissertation Topics

The dissertation covers several different probing algorithms that are appropriate for different types of multiuser channels. To begin, Chapter 2 presents background material on channel models, dynamic channel allocation, distributed power control, and capacity measures. Next, Chapter 3 discusses probing for channels where exact measurements are possible. Two specific examples are discussed: a slow frequencyhopped indoor wireless system and a multitone modulated wireline system. In each case, the channel varies slowly compared to the rate at which the distributed algorithms will operate. This allows precise channel estimation and channel allocation to be performed.

Chapter 4 presents a probing algorithm for systems where the channel cannot be estimated precisely. The example is a wireless cellular system with mobility. Channels are partitioned into power controlled and fixed assignment. Users initially enter the system through a fixed assignment channel to guarantee access. They then probe for a power controlled channel and use power control whenever feasible. The goal is to balance capacity with the limits of time varying channels.

Chapter 5 explores probing for power control with adaptive antennas. The addition of adaptive antennas makes probing much more difficult. Power control and minimum mean square error (MMSE) adaptive antenna arrays are discussed.

Last, Chapter 6 presents concluding remarks on probing and its application to multiuser communication systems.

## Chapter 2

## Background

This chapter introduces background material to lay a technical foundation for the later chapters. It begins with an overview of wireless and wireline channels, including a discussion of current communications systems and the channel impairments they encounter. Next is a review of power control and dynamic channel allocation algorithms. These are new techniques that enhance the performance of multiuser communication systems. The last topic discussed in this chapter is capacity and how it is measured in practical systems.

## 2.1 Wireless Channels

The characteristics of wireless radio channels arise from the physics of radio propagation and vary with the operating environment. In recent years, the mobile radio (cellular telephone) and indoor environments have been studied extensively [28][32][37][42][48][57]. This work has led to a common statistical model that can used to describe the nature of wireless channels.

#### **2.1.1 Propagation Model**

There are three components to the radio propagation model that correspond to three distinct physical mechanisms. The first, path attenuation, arises from the expansion of the electromagnetic wave front as the radio wave travels from the transmitting antenna to the receiving antenna. In free space, the power of the received antenna will be reduced by a factor proportional to the square of the distance between transmitter and receiver. In mobile radio and indoor channels the attenuation can increase at a faster rate, which is expressed as a distance exponent greater than 2. The attenuation slope is increased because the received signal is typically a combination of a direct path and a low angle reflection off the ground [37]. When these signals combine, the resulting signal attenuation is proportional to distance to the fourth power. This effect has been observed in practice, with experimentally determined attenuation exponents between 3 and 5 [32][42]. High attenuation exponents (greater than 4) tend to occur both in dense urban areas and in other areas when the separation between transmitter and receiver is more than 40 km. This additional attenuation is primarily due to signal scattering off objects blocking the signal path. The path attenuation does not vary significantly within the frequency band of a typical radio system.

The second propagation characteristic is shadowing, which includes signal attenuation caused by buildings, hills, wall partitions, or any other object that is very

large compared to the radio wavelength and blocks a line of sight path. Radio propagation can also be enhanced by atmospheric ducting [42], a condition where the atmosphere refracts radio waves towards the earth and improves propagation characteristics over the horizon. The signal path improvement or attenuation is modeled as a log-normal random variable whose variance depends on the environment. As users move, the shadowing varies because of the blocking effect of different objects. Like path attenuation, shadowing does not vary much over the ranges of frequencies employed in most communication systems.

The last propagation characteristic incorporated in the model is fading, a process caused by the combination of many different signal paths at the receiving antenna. The common name for this is multipath and it is modeled as either a Rayleigh or a Rician process. Rayleigh fading occurs when there are a very large number of roughly equal powered, independent signal paths. When the signals combine at the receiver the voltage has a complex Gaussian distribution, as a result of the central limit theorem [41]. The Rayleigh distribution is the amplitude of the complex Gaussian voltage. In terms of signal power, the square of voltage, Rayleigh fading has an exponential distribution.

Rician fading is Rayleigh fading with one large signal component added. This type of fading occurs when the received signal is a combination of a strong direct path and many weaker, scattered paths. The Rician model is more accurate for many situations, but the Rayleigh model is used more often for analysis because it is mathematically simpler and it represents the worst case in signal fading.

Since Rayleigh and Rician fading result from the vector addition of multipath signals, they are a strong function of the phase angles of these signals. Consequently, small changes in frequency or distance vary the phase of the multipath signals and change the fading level. Fading also tends to change with time due to mobility of transmitters, receivers, or other objects in the neighborhood.

The three characteristics can be combined into one statistical expression for the received signal power:

$$_{r} = P_{t}C\left(\frac{1}{d}\right)^{n}X_{f}X_{s} \tag{2.1}$$

where  $P_t$  is the transmitted power, *C* is a constant that incorporates antenna gains and other fixed factors, *d* is the distance from transmitter to receiver, *n* is the attenuation exponent (usually  $n \approx 4$  for mobile radio systems), <sub>*f*</sub> is an exponentially distributed random variable corresponding to the Rayleigh fading, and  $X_S$  is the shadowing random variable:

$$X_s = 10^{\left(\frac{x}{10}\right)} \tag{2.2}$$

where x has a Gaussian distribution, typically with standard deviation  $\sigma = 8$  decibels.

#### 2.1.2 Cellular Systems

The cellular concept was developed at Bell Laboratories in the 1970s as part of the Advanced Mobile Phone System (AMPS). In this system, the coverage area is divided into hexagonal cells and a base station is located at the center of each cell. Users in a given cell only communicate with that cell's base station, as shown in Figure 2-1. However, they also generate interference to nearby users that share the same channel.To mitigate interference, channels are assigned to each cell according to a re-use pattern. For example, a common pattern is K=7, in which channels are re-used every seven cells. This forces the closest interferer to be at least two cells away.

Figure 2-1 Cellular Concept with Re-use Factor 7



Under worst cast conditions, the signal-to-interference ratio (SIR) on the desired user's link will be fixed by the ratio of the distances  $R_0$  and  $R_1$ , if shadowing and fading are neglected. The SIR,  $\gamma$ , is then lower bounded:

$$\gamma \ge \left(\frac{R_1}{R_0}\right)^n \tag{2.3}$$

where n is the same propagation loss exponent given in Equation 2.1. Re-use factors and associated distance ratios are given in [37].

The SIR level on a channel is critical because it determines the transmission rate that can be achieved. More sophisticated channel access algorithms all work to maintain an SIR level that is high enough for each user to communicate, but no higher than necessary. The goal is to adapt to actual interference conditions rather that relying on worst case channel allocations. In this way, capacity can be increased by reducing the average re-use distance for channels.

### 2.2 Wireline Channels

Wireline systems have many physical forms including coaxial cable, waveguide, twisted pairs. In this work, twisted pairs in a cable bundle are studied because this is the medium for future multiuser wireline communication systems, such as the asymmetric digital subscriber line (ADSL) [55]. The ADSL system was developed to allow high speed digital communication to homes over traditional telephone wiring. The system employs very fast data rates, 6 megabits per second (Mbps) from the telephone company central office to the home and 768 kilobits per second (Kbps) in the reverse direction. Applications include cable television, internet access, and videophone service.

#### 2.2.1 Twisted Pair Cable Characteristics

Twisted pair cables have three major channel impairments. These are

attenuation with distance, near end cross talk (NEXT), and far end cross talk (FEXT). Each of these is frequency dependent and can be modelled as a frequency response. The attenuation of the twisted pair channel,  $G_0(f)$ , is determined by the cable length and by the presence of bridge taps at different points along its length.  $G_0(f)$  can be determined by an equivalent circuit model for transmission lines. For the results in the dissertation, the frequency response models for Bellcore [5] standard loops are used.

The NEXT and FEXT terms are computed with a model based on [38]. NEXT has a fixed frequency response:

$$G_{NEXT}(f) = \left(\frac{K_N}{10}\right)^{10} \left(\frac{f}{f_0}\right)^{3/2}$$
(2.4)

The cut-off frequency for NEXT,  $f_0$ , is 0.772 MHz. FEXT is also affected by the frequency response of the disturbed twisted pair and the length of the cable, *l*:

$$G_{FEXT}(f) = \left(\frac{K_F}{10}\right)^{10} \left(\frac{f}{f_0}\right)^2 \left(\frac{l}{l_0}\right) G_0(f)$$
(2.5)

In this equation,  $f_0 = 3.15$  MHz and  $l_0 = 1000$  feet.  $K_N$  and  $K_F$  are gamma distributed random variables with mean and variance determined by the type of cable [38]. The gamma distribution is similar to the Gaussian distribution in that it can be completely determined by its mean and variance, however it has a different structure in the tails. The gamma distribution is:

$$P[x \le x_0] = \frac{\beta^{\nu}}{\Gamma(\nu)} \int_0^{x_0} x^{\nu - 1} e^{-\beta x} dx$$
(2.6)

with parameters  $\nu$  and  $\beta$  derived from the mean and variance with the equations:

$$\nu = \left(\frac{m}{\sigma}\right)^2 \qquad \beta = \frac{m}{\sigma^2}$$
 (2.7)

and the gamma function is a generalized factorial based on Euler's integral:

$$\Gamma(\mathbf{v}) = \int_{0}^{\infty} t^{\mathbf{v}-1} e^{-t} dt , \, \mathbf{v} > 0$$
 (2.8)

#### **2.2.2 Wireline Signaling Format**

It is assumed that a multiple signal format such as that proposed in the ADSL standard is used on the wireline channel. The ADSL system uses discrete multitone transmission (DMT) [7][15]. This modulation format partitions the frequency spectrum into bins that are 4.3125 KHz wide with a discrete Fourier transform. The frequency partitioning allows each bin to be modulated separately with a constellation size that is appropriate for the signal-to-interference ratio that can be achieved on the bin. In the downstream link (from service provider to residence), there are 256 frequency bins. A plot of typical gain, NEXT, and FEXT, between twisted pairs in a PIC cable on the downstream path is given in Figure 2-2. Like the wireless channel, the ADSL wireline channel becomes interference limited when the cable bundle is shared by multiple user pairs.

Figure 2-2 Loop Gain with Typical NEXT and FEXT Response



#### 2.2.3 Wireline Channel Access

Current ADSL systems employ a multiple access technique that is analogous to fixed frequency re-use in the cellular wireless system. Each user enters the system at a fixed power level and allocates a QAM constellation for each frequency bin based on the received signal level. The interference level is assumed to be the 1% worst case for a congested cable [55]. In practice, however, the maximum SIR level will depend on the specific cable characteristics and the power levels of other users on the cable. System capacity, in terms of the average user transmission rate, can be improved by adapting to the actual interference conditions. Section 3.4 explores the capacity improvement that is possible with probing and dynamic power allocation.

## 2.3 Multiuser Channels

From a multiuser standpoint, wireline and wireless channels are fundamentally similar as shown in Figure 2-3 and Figure 2-4. Consequently, signalling techniques that have been applied to wireless systems can also be applied to wireline systems. Both systems are called multiuser because any user's transmission will cause interference to other users. To maximize performance, the transmitter signaling and power level must be coordinated between every user in the system.









A general model for multiuser channels is the interference channel shown in Figure 2-5. The model consists of N coupled transmitter-receiver pairs. Each pair represents a link, such as user-to-base station for a cellular system, or user-to-user in a wireline twisted pair. Each link receives interference from every other link, and from thermal noise in the receiver. The ( $N \times N$ ) link matrix, G, is defined by the power transfer functions between each transmitter and each receiver. As shown in Figure 2-5, element  $G_{ij}$  refers to the gain from the transmitter on link j to the receiver on link i.

Figure 2-5 Interference Channel Model



The links have signal-to-interference ratios:

$$_{i} = \frac{P_{i}G_{ii}}{\sum_{j \neq i} P_{j}G_{ij} + N_{i}} \text{ for link } i = 0...N - 1$$
(2.9)

where  $P_i$  is the power of the transmitter on link *i* and  $N_i$  is the power of the thermal noise.

In most systems, the interference channel will be divided into independent subchannels or frequency bins. Each of these is considered to be a separate communications channel. A link may or may not see the same interferers on each subchannel. The entire interference channel is characterized by a transfer function matrix and a noise vector for each sub-channel.

There are two major differences between wireless and wireline channels. First, the gain matrix, G, for wireless channels tend to vary much more quickly than wireline channels. Second, the amount of coupling between users, characterized by the off diagonal elements of G tend to be much smaller for wireline users. These difference affect both probing and bit allocation strategies.

## 2.4 Multiuser Channel Capacity

The concept of channel capacity was introduced by Shannon [50] as the definitive upper bound on the transmission rate over a communication channel. For Gaussian channels, it is easy to compute the capacity [17]:

$$C = W \log \left( 1 + \frac{P}{N_0 W} \right) \tag{2.10}$$

where P is the signal power, W is the bandwidth, and  $N_0$  is the noise power per unit bandwidth. Recent developments in modulation and coding have demonstrated that it is possible to achieve rates that approach capacity on Gaussian channels. These methods are used in the telephone line modems we use every day.

The situation is more complicated for multiuser channels. First, for a multiuser channel there is not a capacity number, but rather a region of achievable rates with as many dimensions as users sharing the channel. Second, it is very difficult to compute the capacity of multiuser channels. In fact, the capacity of the interference channel (our system model) is unknown [17], although there has been significant progress in determining appropriate bounds [9][10][12][26]. At the present time, however, a direct evaluation of multiuser system performance with theoretical capacity is not possible.

#### **2.4.1 Practical Restrictions on Signaling Format**

The alternative to comparing a multiuser signaling algorithm to theoretical capacity is to compare new signaling algorithms to older ones. To make these comparisons fair, a set of practical capacity measures must be used that take into account the limitations of the signaling and detection methods used in real systems.

For wireless systems, it is assumed that some type of time and frequency division multiple access is employed. For frequency division multiple access (FDMA), the spectrum is divided up into orthogonal sub-channels on different frequencies. Each of these sub-channels is an independent interference channel. Since different subchannels may have different propagation characteristics, there is often a benefit in allowing each user to use each channel. This is accomplished with slow frequency hopping [51].

For ADSL wireline systems, it is assumed that discrete multitone transmission (DMT) is used. DMT divides the channel into 256 sub-channels on the downstream path (service provider to user) and 32 sub-channels on the upstream path (user to service provider). Every user shares every sub-channel and no time division is used.

Quadrature Amplitude Modulation (QAM) is the standard modulation format, and the maximum number of bits per symbol is:

$$b = \log\left(1 + \frac{\gamma}{\Gamma_{gap}}\right) \tag{2.11}$$

where  $\gamma$  is the signal-to-interference (SIR) level and  $\Gamma_{gap}$  is the gap between the system performance (required SIR for a desired bit rate) and the theoretical minimum SIR. In all cases it is assumed that a single user detector is employed and that the interference appears as Gaussian noise.

#### **2.4.2 Practical Capacity of Interference Channels**

For wireless cellular systems, one performance measure is the user capacity, which is defined as the offered load (in Erlangs) per cell that will yield a blocking probability of 1%. The assumptions are that user arrivals are random, independent, and uniformly distributed in space across the system. The arrival process is Poisson and the user hold times are exponentially distributed. Comparisons between systems assume that the number of channels (system bandwidth) is the same.

For wireline systems, capacity is measured with a revenue function for the system and algorithms are compared in terms of revenue. A typical revenue function has the form:

$$_{a} = \frac{1}{N} \sum_{i} N_{i} R_{i} - \alpha N_{d}$$
(2.12)

where N is the total number of users admitted to the system,  $N_i$  is the number of users admitted at rate class *i* with corresponding revenue  $R_i$ .  $N_d$  is the number of users dropped (forced out of the system by new users) and  $\alpha$  is the weighting factor for dropping.

## 2.5 Dynamic Channel Allocation

Since the modulation and detection methods are fixed, wireless system capacity will be determined by the allocation of channels and power to users. A baseline system is the cellular system with fixed frequency re-use. Here channels are allocated by fixed patterns according to worst case interference conditions. Capacity can be increased by allocating channels and power to users in a more efficient manner. This process is called dynamic channel allocation (DCA) because channels are allocated to users as they need them.

In the fixed re-use cellular system, channel access is determined at the cell level and can be easily determined by the base station. If the channel is not currently in use, then it can be allocated to a user. With DCA, the assignment of channels to cells (and to users) is no longer fixed. The advantage of DCA is that it can employ an adaptive reuse distance for each channel. Since most users are not operating under worst case conditions they can often tolerate a smaller re-use distance. Over the set of all users, the average re-use distance is reduced. This increases the number of users per channel, a common and important measure of system capacity.

The disadvantage of DCA is an increase in system complexity. Since channels are not directly allocated to cells, channel access can no longer be determined at the cell level. Sophisticated allocation techniques require a new user to be able to determine whether or not it is feasible to access a channel. Channel probing provides this information and consequently, DCA and probing must work together. In general, the probing algorithm is first applied to determine the channel characteristics. Next, a choice of channel allocation is made based on which channel (or group of channels) can best carry the required allocation of bits. Power control is then used to allocate the proper transmitter power to each user so that every SIR constraint can be met at all times.

## 2.6 Power Control

#### **2.6.1 Fundamental Concepts**

The objective of power control is to compute a power level for every user in order to meet a set of SIR constraints. This problem has been explored in [69]. In the general problem, individual users will have users different SIR constraints. In an N user

system the vector of SIR constraints for the users is  $\gamma = \{\gamma_0, \gamma_1, ..., \gamma_{N-1}\}$  and the

vector of user transmitter powers is  $p = \{p_0, p_1, ..., p_{N-1}\}$ . Define a matrix H with

off-diagonal elements  $(i \neq j) h_{ij} = -\frac{\gamma_i G_{ij}}{G_{ii}}$  and diagonal elements (i = j) equal to 1.

Also, define a noise vector *n* with elements  $n_i = \frac{\gamma_i N_0}{G_{ii}}$ . This term represents the

AWGN in the receiver scaled by the gain on the desired path and the SIR constraint. The set of feasible solutions has the constraint:

$$Hp \ge n \tag{2.13}$$

and the optimal solution (in terms of minimum power) is found when equality holds. For a given gain matrix *G* and SIR requirement set  $\gamma$ , a feasible solution exists when:

$$det(H) > 0 \tag{2.14}$$

and all principal minors of H are greater than zero. Under this condition, the all positive optimal power vector is:

$$p^* = H^{-1}n \tag{2.15}$$

### 2.6.2 Distributed Power Control

In a large system it is desirable to compute  $p^*$  in a distributed fashion. This has been explored in [3][4][18]. In [4], the set of users is partitioned into an active group that maintain their SIRs by a power update and a new-user group that increase their
powers in fixed steps until they either achieve the desired SIR level or give up. The power update equation for active users is:

$$p_i(n+1) = \frac{p_i(n)\delta\gamma_i}{r_i(n)}$$
(2.16)

where  $r_i(n)$  is the SIR measured by user *i* at time *n*,  $\gamma_i$  is the target SIR level and  $\delta$  is a small constant greater than 1. New users initiate transmission at a very low power level and increase their powers in fixed steps:

$$p_i(n+1) = \delta p_i(n) \tag{2.17}$$

Under this distributed algorithm, the SIRs of active users will be maintained between  $\gamma$  and  $\delta\gamma$  and new user SIR levels will converge to a maximum feasible SIR level as *n* becomes large.

### 2.7 Summary

Future multiuser systems will incorporate both DCA and power control to help maximize capacity. (The acronym DPCA is often used for the combination of power control and dynamic channel allocation.) Recent work [65] in DPCA algorithms has yielded very high capacity algorithms that offer significant improvements over fixed allocation techniques. However, these algorithms require precise knowledge of the achievable SIR level for each user. They also require precise power control to insure that every user can maintain its SIR level. In a complex communication system, such as a cellular or a PCS network, it would be difficult for a central controller to measure and maintain all the information necessary to implement the next generation of DPCA algorithms. The fact that these algorithms work best when each user employs frequency hopping (or another method to access multiple channels) makes the implementation problem even more difficult.

The solution to the complexity problem lies in distributed algorithms. The distributed power control algorithm mentioned above is well understood and can achieve virtually the same accuracy as centralized power control in a wide variety of environments. The next step is to make the entire channel allocation and user admission process distributed and yet retain the capacity of the new DPCA algorithms. This can be achieved with channel probing.

Channel probing assumes that each active user in a multiuser system will employ distributed power control to maintain its SIR level. New users probe each channel at low power to determine the maximum feasible SIR level that they can achieve without disturbing the active users. This information is then applied to a DPCA algorithm. Once a channel (or set of channels) is selected, the new user admits itself directly to the network. No centralized control is needed.

Research in channel probing consists of developing algorithms and exploring how well they work. The algorithms discussed in the next three chapters have been tailored for different wireless and wireline systems. In some systems it is possible to accurately measure signal and interference power levels while in others it is not. This influences both the channel allocation algorithm and also the probing method. When it is possible to precisely measure the channel it is also possible to probe very accurately and achieve a very well engineered system. When accurate measurements are not possible, the probing algorithm will use a heuristic that tries to balance the goals of rapid response and good admission decisions.

## **Chapter 3**

# **Probing Power Controlled Channels**

This chapter explores the fundamental algorithm for channel probing and the limitations on its accuracy. The emphasis is on communication systems where the channel is very stable relative to the symbol rate. In these systems, the gain matrix, G, will remain constant long enough for users to estimate the important parameters and choose a channel and power allocation. Stable channels include indoor wireless channels and ADSL wireline channels. For these channels, it is possible to perform precise bit allocation using the channel probing information.

## **3.1 Probing for Maximum Feasible SIR**

In Section 2.6 the concept of a feasible set of SIR constraints was introduced. For distributed power control to operate properly, each user must pursue an SIR constraint,  $\gamma_i$ , that is feasible. To simplify the choice of SIR level, each user is forced to select a level at admission time. It is assumed that the channel is currently occupied by other users that are operating with distributed power control. When a user enters the system it is only allowed to select an SIR level that is less than the maximum feasible level, given the other user SIR constraints. This way the active users on the channel are always driving the distributed power control to a feasible solution.

The maximum feasible SIR level that a new user can obtain on a power controlled channel is:

$$\gamma_{max} = \frac{G_{ii}(p_1 - p_0)}{q(p_1) - q(p_0)} \tag{3.1}$$

where  $p_1$  and  $p_0$  are two different transmitter powers,  $p_1 > p_0$ ,  $G_{ii}$  is the gain on the signal path, and  $q(p_1)$  and  $q(p_0)$  are interference measurements made while transmitter power levels  $p_1$  and  $p_0$ . This result was first approximated in [27] and later shown to be exactly true in [65].

The initial power level,  $p_0$ , can be set to zero and the probing result still holds. In fact, with this information, the new user can also compute r(p), its SIR level at transmitter power p:

$$r(p) = \frac{pG_{ii}\gamma_{max}}{q(0)\gamma_{max} + pG_{ii}}$$
(3.2)

The probing solution for r(p) is identical to what the new user would achieve by increasing its power according to equation (2.17) to p and then waiting for the distributed power control algorithm to converge completely. However, the probing solution offers several advantages. First, probing can be performed much more quickly. This is important if several channels must be probed to determine which is the best for bit allocation. Second, the probing can be performed at a very low power. This prevents excessive interference to active users if the channel turns out to be very congested.

## **3.2 Probing Accuracy**

The accuracy of channel probing is critical because it places a fundamental limit on the performance of any DPCA algorithm that uses the probing results. When the probing result is in error, the DPCA algorithm may choose the wrong channel or allocate the wrong constellation size to a channel. This results in a loss in capacity relative to a system where perfect measurements are possible. In a dynamic environment, probing errors are inevitable even if the measurement technique is optimal. This section explores the sources of probing errors and discusses ways to limit them in practice.

Channel probing depends on both the accuracy of signal and interference measurements and on the accuracy of the underlying channel model. It is assumed that the channel varies extremely slowly relative to the symbol rate of the system. This assumption is necessary in order to measure received power, received interference, and the channel gain. If the channel varies on the order of the symbol rate, such as in a low data rate mobile system with rapid Rayleigh fading, distributed power control is only possible with averaging over the fading. In this case a different power control strategy is needed and precise channel probing is not feasible.

It is also assumed that the channel gains vary slowly with respect to the power control update rate of the system. This assumption is necessary to insure convergence of the distributed power control. The results in Section 3.2.1 and Section 3.2.2 depend on exact convergence. In mobile radio systems, however, shadowing and other effects can limit the ability of the power control to track the channel. When this occurs the effectiveness of channel probing becomes limited and a different strategy must be employed. A method for implementing DPCA under these conditions is discussed in Chapter 4.

#### **3.2.1** Estimation With Coherent Detection

A coherent receiver allows the probing user to make accurate measurements of the maximum feasible SIR on the power controlled channel. The system model in Figure 3-1 is assumed.

Automatic gain control (AGC) is employed to compensate for the transfer function of the channel. A known probing sequence is transmitted to the receiver or, if the SIR is high enough, a decision based loop is used to adjust the AGC. In either case, it is assumed that the knowledge of the channel gain is perfect and gain errors will not be introduced into the estimate of the interference power. To compute the estimate of  $\gamma_{max}$ , the transmitter sends a probing signal at the initial power level. The receiver waits until the interference power on the channel has settled and then begins the measurement of its first interference power estimate,  $\hat{q_1}$ . The interference power is averaged over a frame of length *N* and stored.



Figure 3-1 Interference Power Estimation

The transmitter then increases its power by a factor  $\delta$ . Once again the receiver waits until the interference power on the channel has settled and then it measures the second estimate,  $\hat{q}_2$ . The received signal powers are computed from the AGC settings, yielding  $s_1$  and  $s_2$ . The maximum feasible SIR estimate is computed with:

$$\hat{\gamma}_{max} = \frac{s_2 - s_1}{\hat{q}_2 - \hat{q}_1} \tag{3.3}$$

To simplify the error analysis, the inverse of  $\hat{\gamma}_{max}$  is used. Let  $B = \gamma_{max}^{-1}$  and let  $\hat{B}$  be

the estimate of B. Then,

$$\hat{B} = \left(\frac{1}{(\delta - 1)s_1}\right)(\hat{q}_2 - \hat{q}_1)$$
(3.4)

If the interference seen at the receiver is Gaussian, then the interference estimates and  $\hat{B}$  will be chi-square distributed [45]. Using this fact:

$$E[\hat{B}] = B \tag{3.5}$$

$$Var[\hat{B}] = \frac{2B^2}{N} \left[ 1 + \frac{2q_2q_1}{(q_2 - q_1)^2} \right]$$
(3.6)

Therefore, the estimate of *B* is unbiased and depends on the frame length used for the estimates and the change in interference level between the two estimates. Since the estimate is unbiased, the variance is the same as the mean-square error (MSE). Making the substitution,  $q_2 = (1 + \varepsilon)q_1$  the normalized root-mean square (RMS) error is:

$$Error = \sqrt{\frac{2}{N} \left(1 + 2\frac{(1+\varepsilon)}{\varepsilon^2}\right)}$$
(3.7)

The error in the estimate of the maximum feasible SIR is due to the size of the frame used to make the estimate and the gap in interference level between two measurements. This fact can be exploited to allow new users to know how accurate their estimates of  $\gamma_{max}$  will be. After making an initial measurement of the interference power, the new user can increase its transmitter power until it measures a change in interference that is large enough for an accurate estimate. If, after increasing the probing power by several decibels, the interference does not increase, the new user will

know that any practical SIR level will be feasible.

#### **3.2.2 Estimation Without Detection**

Sometimes it is desirable to probe the channel at a very low SIR level. For a low power probe, it might be impossible to provide carrier synchronization, symbol synchronization, or accurate decisions. However, it is still possible to probe the channel with a series of power measurements. It is assumed that a square law detector is used to measure power, and the results are averaged over multiple symbols to improve the accuracy. Since there is no symbol detector to separate the received signal from the interference, it is necessary to provide a way for the receiver to make several different measurements. It is also assumed that the system will allow the probing user to measure the receiver noise, desired signal plus receiver noise, and the interfering signal plus receiver noise on separate channels or segments. This is possible if synchronization information is made available on a setup channel.

The receiver noise is averaged over  $N_1$  symbols and stored in  $r_1$ . The transmitter turns on its higher power setting and the receiver averages  $N_1$  symbols and stores the result in  $r_2$ . As before, the new user measures interference levels  $\hat{q_1}$  and  $\hat{q_2}$  by averaging over  $N_2$  symbols. The received signal power is estimated with:

$$\hat{s}_2 = r_2 - r_1 \tag{3.8}$$

and the inverse of the maximum signal to interference ratio is:

$$\hat{B} = \left(\frac{\delta}{(\delta-1)\hat{s}_2}\right)(\hat{q}_2 - \hat{q}_1) \tag{3.9}$$

This is the same as equation (3.4), except that the larger signal power measurement is used because it will have a smaller error. The estimate of the signal power is unbiased  $(E[\hat{s}_2] = s_2)$  and its variance is a function of the receiver noise power,  $\sigma_n^2$ , and the number of symbols used in the estimate:

$$Var[\hat{s}_{2}] = \frac{4}{N_{1}} s_{2} \sigma_{n}^{2} + \frac{4}{N_{1}} \sigma_{n}^{4}$$
(3.10)

This holds true for any zero mean signal that is not correlated with the receiver noise. To compute the statistics of the estimate  $\hat{B}$  the approximation from [41] is employed:

$$E[g(x)] \cong g(\eta_x) + g''(\eta_x) \frac{\sigma_x^2}{2}$$
(3.11)

where  $\eta_x$  and  $\sigma_x^2$  are the mean and variance of *x*. The expected value of the probing estimate is:

$$E[\hat{B}] \cong B\left(1 + \frac{4\sigma_n^2}{s_2N_1} + \frac{4\sigma_n^4}{s_2^2N_1}\right)$$
(3.12)

which shows that the estimate is asymptotically unbiased. An approximation of the RMS error is:

$$rror \cong \sqrt{\frac{2J}{N_2} + \frac{4\sigma_n^2}{s_2N_1} + \frac{4\sigma_n^4}{s_2^2N_1^2} + \frac{24J\sigma_n^2}{s_2N_1N_2} + \frac{24J\sigma_n^4}{s_2^2N_1^2N_2}}$$
(3.13)

where  $J = 1 + 2\frac{(1+\epsilon)}{\epsilon^2}$  and  $\epsilon = \frac{q_2}{q_1} - 1$  as in the error estimate for the coherent

detection.

To compare the two probing methods, assume that a probing segment is used at the beginning of each frame and it occupies 10% of the frame length. Also assume that the signal to noise ratio (SNR) for the signal is 0 dB. Substituting  $N_2 = 0.9N$ ,

$$N_1 = 0.1N$$
, and  $\frac{\sigma_n^2}{s_2} = 1$  into equation (3.13) yields:

$$rror = \sqrt{\frac{2.222J + 40}{N} + \frac{266.7J + 400}{N^2} + \frac{2667J}{N^3}}$$
(3.14)

A plot of the estimation error in dB for the two probing techniques is given in Figure 3-2. The top two curves show the error for a change in interference level of 0.25 dB and the lower two curves show the error for a change in interference level of 2.0 dB. From these curves it can be seen that for N > 100 the difference between the two probing techniques becomes small compared to the error. This implies that estimating the channel gain without symbol detection does not introduce a significant error if N can be made sufficiently large. For  $N = 10^4$  it is possible to estimate the maximum feasible SIR within 1.5 dB. Thus, for stable channels it is possible to probe the power controlled channel very accurately.

Figure 3-2 Probing Error Due to Interference Estimation



#### **3.2.3 Effect of Distributed Power Control**

Another source of error for channel probing is the convergence of distributed power control. Channel probing requires precise measurements of interference from the active users. If the interference levels are changing because of the distributed power control algorithm, this will lead to errors in the estimate of  $\gamma_{max}$ . To completely avoid these errors, the probing user must wait for the interference level to converge completely. Since this may not be practical, a bound on the error due to power control convergence is derived.

From [4] we know that the distributed power control will converge whenever

there exists a feasible solution for equation (2.13). However, the rate of convergence depends on matrix H. The distributed power control algorithm is equivalent to a matrix update equation for the power vector:

$$p(k+1) = Fp(k) + n$$
(3.15)

where

$$F = I - H \tag{3.16}$$

and *I* is the  $N \times N$  identity matrix. When a new user probes the channel the active users see a change in the background interference level and adjust their powers upward to compensate. The interference seen by the new user is the inner product of p(k) and a vector of transfer functions from the active users to the new receiver,  $g_n = [g_{n0}, g_{n1}, ..., g_{n(N-1)}]^T$ . Consequently, the interference level follows the convergence rate of the current set of active users. When probing we may assume that the active users have converged completely for the first interference measurement. The new user then turns on its transmitter power and waits *m* steps before measuring the second interference level. The difference in the interference measurements is:

$$q_2(m) - q_1 = g_n^T [I - F^m] (p_2 - p_1)$$
(3.17)

and as *m* becomes large,  $F^m \rightarrow 0$  and the difference will increase uniformly to the desired result:

$$g_n^T(p_2 - p_1) (3.18)$$

Using the Holder inequality [30], the absolute error at step m can be bounded by:

$$rror \le \|g_n\|_{\infty} \|F\|_1^m \|p_2 - p_1\|_1 \tag{3.19}$$

In many systems, the change in interference will be dominated by the change in power of the strongest interferer. We can then make the substitution:

$$\|g_n\|_{\infty} \|p_2 - p_1\|_1 = Cg_n^T(p_2 - p_1)$$
(3.20)

where *C* is a small constant greater than 1. Then, the normalized error is bounded by  $C \|F\|_1^m$ . As *m* becomes large, the error can also be bounded by the spectral radius (largest eigenvalue) of *F*, however this result is less useful for probing since the goal is to keep *m* as small as possible.

The error result has several implications. Unlike the errors due to estimation of interference power levels, the error due to power control convergence cannot be reduced by increasing the difference in transmitter power between each probing step. Also, as active users choose their SIR targets closer to the maximum feasible, the norm of F will increase, convergence will slow down, and accurate probing will take longer.

To measure the probing error in due to convergence, a 19 cell wireless system with multiple channels and slow frequency hopping was simulated. We assumed that interference measurements are averaged over frames long enough to make the measurement error insignificant. Consequently, any errors due to interference estimation are ignored. Distributed power control is used and probing users wait from 2 to 6 steps for the power control to converge before making measurements. The measurement procedure is as follows. A new user enters the system and probes 23 channels at low power. It then stores those results and slowly increases its power according to equation (2.17) for 75 steps. It then compares its current SIR level on each channel with the probing estimates. The results are given in Figure 3-3.

Figure 3-3 Probing Error from Power Control Convergence



The probability of a probing error greater than 2 dB is plotted as a function of the number of steps, m. For m > 5 the error rate no longer decreases, indicating that additional convergence time does not improve the probing estimate significantly. Since error due to power control convergence must decrease with increasing iterations, the residual error is not due to power control convergence. Therefore, we conclude that the

convergence error is negligible for m > 5.

#### **3.2.4 Effect of User Dynamics**

The last probing error we investigate is the effect of user dynamics, the entry and exit of users from the system during probing. There are several specific errors that fall into this category. For example, two or more users may probe the channel at the same time causing inaccurate estimates of  $\gamma_{max}$ , or, a user might exit the system while a new user is making interference measurements. In the latter case, the interference may actually drop during probing which leads to a completely erroneous result. In general, errors of this type can be reduced or eliminated if probing can be done quickly relative to the dynamics of the system. Of course, any probing result will be out of date once an active user leaves the system. However, if probing is used as part of an admission control procedure, active users will be able to maintain the SIR levels and bit allocations they determine from probing.

To measure the probing error due to user dynamics, the simulation program described in Section 3.2.3 is used. The new user waits m = 5 iterations after probing to eliminate convergence errors. The resulting probing error is then assumed to be entirely due to user dynamics. The new user stores the probing results and slowly increases its power according to equation (2.17) for 75 steps. It then compares its current SIR level on each channel with the probing estimates.

The entire interval from initial probing to final SIR measurement is called the

probing interval. After the probing interval, the new user is admitted to the system as an active user, where it maintains target SIR levels on a subset of the channels with distributed power control. The active time in the network is an exponentially distributed random variable. The average active time was varied over several simulation runs to evaluate the effects of user dynamics. Figure 3-4 presents a plot of the simulation results.



Figure 3-4 Channel Probing Performance

As the ratio of probing time to active time approaches 1%, the user will be able to probe within 3 dB of the correct SIR level about 85% of the time and within 1 dB about 80% of the time. An error of 3 dB is significant, because that corresponds to the power change to allocate an additional bit for systems that employ multiple constellation sizes. When the probing errors are below this amount we know the system is accurate enough to make bit allocation decisions as part of an admission algorithm.

# **3.3 Application of Probing to Indoor Wireless** Systems

Sophisticated DPCA algorithms for indoor wireless applications have been developed based on adaptive bit allocation [65]. Of these algorithms, the best performing is a minimum hop interference avoidance strategy. With this strategy, each user must allocate a set of M bits over a group of N frequency hops with the minimum number of hops possible. If a feasible allocation cannot be found, the user is blocked.

The minimum hop strategy achieves high capacity by allowing users to adapt to the actual interference conditions. Users near the base station can use a few slots at high SIR levels and achieve very high re-use factors. At the same time, users at the cell edge. can use more channels at a lower SIR level. Power control limits the power of each user to the minimum necessary for communication.

#### **3.3.1** Simulation of System with Probing

The results in [65] were based on a centralized implementation of the interference avoidance DPCA algorithm. We present new results to demonstrate that this algorithm can be implemented in distributed fashion with channel probing. The capacity loss in the distributed version of the algorithm is caused by probing errors and

power control inefficiencies that occur with distributed power control (DPC). The simulation helps to quantify this capacity loss. The specifications of the system we simulated are given in Table 3-1.

Channels	23
Constellation sizes	0, 2, 3, 4, 5, or 6 bits/ symbol
	5,11001
Γ (SIR gap)	4 dB
Average bits/symbol	2.17 (50 bits / 23 hops)
Propagation Loss	R <sup>-3</sup>
Shadowing	10 dB standard devia-
	tion
Number of Cells Simu- lated	19

**Table 3-1: Frequency Hopped System with Probing** 

The frequency hopping patterns for each user are determined by a group coincidence system [65] in which a user sees interference from the same set of users in each hop. Interference is avoided by the choice of channels that each user makes at admission time. New users admit themselves with the following algorithm:

- 1. If no hopping pattern is available, block user. Otherwise continue.
- 2. Select a hopping pattern that is free and has not been probed previously.
- 3. Probe all hops to determine  $\gamma_i$  and  $b_i$  terms.
- 4. If total number of bits < 60 (50 + 10 bit safety margin) mark this hopping

pattern as unusable and go back to step 1. Otherwise continue.

- 5. Allocate bits to channels starting from highest to lowest  $\gamma_i$ .
- 6. Follow power update equation (2.17) for each channel until SIR level is achieved.
- 7. Admit user to the system and use DPC equation (2.16) to maintain SIR level.

#### **3.3.2 Simulation Results**

Figure 3-5 is a plot of the blocking probability of the distributed DPCA system with channel probing as a function of Erlang load. Two different simulation runs are included. The fast run has probing times 1% of the user active time and the slow run has probing times of 0.06% of the user active times. These correspond to the operation when probing is very accurate. For comparison, the performance of the centralized algorithm is included in the plot. The fast run has 84% of the Erlang capacity of the centralized system while the slow run has 91%.

The capacity of the distributed system is less than the centralized system because errors force the distributed system to test for more bits (the safety margin) than necessary to insure that it can be admitted to the system. However, this loss is small compared to the gains from dynamic channel allocation and power control. A fixed channel system (without frequency hopping) with the same bit load and number of channels would have about 1/5 the Erlang capacity of the distributed system.

Figure 3-5 Performance of Indoor Wireless Systems



### **3.4** Application of Probing to ADSL Wireline Systems

In the indoor wireless system explored in Section 3.3 the goal of the access algorithm was to admit as many users as possible, with each user sending data at the same rate. For the ADSL system, however, the goal is to maximize the average bit rate or a cost (revenue) function over bit rates of users that occupy the channel. The number of links that can be supported is fixed by the number of twisted pairs in the cable bundle. However, users can be offered different bit rates depending on the transmission characteristics of the channel.

#### **3.4.1** Distributed Access for Wireline Systems

The distributed access method for the wireline system is very similar to its wireless counterpart. A new user first probes the channel for the maximum possible SIR level and bit rate on each frequency bin. Next, it determines the maximum rate class that it can achieve from the total number of bits it can allocate. It then chooses a bit allocation that will achieve that rate class, increases the transmitter power on each bin to the necessary level and initiates data transmission. Once active, it maintains the SIR levels on each frequency bin through distributed power control. The different rates are shown in Table 3-2. Rates 1 through 4 are defined in the ADSL standard [55] and rates 5 through 7 were added for cases where significant improvement is possible with probing.

Class	Upstream Rate (Kbps)	Downstream Rate (Kbps)
1	192	1536
2	384	3072
3	576	4608
4	768	6144
5	960	7680
6	1152	9216
7	1344	10752

 Table 3-2: ADSL Rate Table

With probing and power control, active users always maintain their SIR levels and consequently dropping does not occur. Therefore, the system revenue can be simplified from equation (2.12) to:

$$R_a = \frac{1}{N} \sum_i N_i R_i \tag{3.21}$$

where N is the total number of users,  $N_i$  is the number of users admitted at rate class i with corresponding revenue  $R_i$ . Revenue can be set to equal the bit rate of the class. Revenue is maximized by allowing each user to transmit data at the highest rate possible, subject to the constraints of interference at admission time. Like the bit allocation problems explored in [65], an optimal revenue maximizing algorithm would require an exhaustive search over the space of feasible bit allocations. Since a search of this type is not practical, the admissions based algorithm seeks to maximize the aggregate revenue by each user maximizing its own bit rate, subject to the current channel constraints.

#### **3.4.2 ADSL Simulation**

A simulation program was developed to compare the performance of the probing system against the present ADSL system. As mentioned in Section 2.2.3, the present ADSL system uses a form of fixed allocation. Users allocate a constellation size to each bin according to the received signal level on that bin and an assumed fixed interference level. The interference level represents the 1% worst case cross talk level based on measurements of real cables [38].

The probing algorithm used in the ADSL simulation is the same as for the wireless simulation. The bit allocation algorithm is slightly different, however. In the

wireless system, the goal was to minimize the total number of channels that were used for the bit allocation. For the wireline system, it is assumed that every user will allocate bits on every available channel. Therefore, the goal is to minimize the excess interference on each channel. The bit allocation starts by allocating as many bits as possible and then removes bits starting with the channel with the least SIR margin until the target number of bits has been reached. The complete algorithm is:

- 1. Probe all frequency bins to determine  $\gamma_i$  and  $b_i$  terms.
- 2. Total the  $b_i$  terms to compute the maximum feasible rate class.
- Sort frequency bins according to excess SIR over that needed for allocated constellations.
- Repeat step 5 until the total number of allocated bits equals that necessary for the rate class in Step 2.
- Decrease the bit allocation in the frequency bin with the smallest excess SIR by one bit.
- 6. Follow power update equation (2.17) for each bin until SIR level is achieved.
- 7. Admit user to the system and use DPC equation (2.16) to maintain SIR level.

The simulation system parameters follow the ADSL standard as closely as

possible. A complete listing of the system parameters is given in Table 3-3.

Duplexing method:	Frequency division
Upstream bins allocated:	7 through 31
Downstream bins allo- cated:	32 through 255

**Table 3-3: Simulation Parameters** 

Bits allocated per bin:	2-15, or zero
Bit allocation SIR margin:	12 dB (8 dB + 4dB coding gain)
	6 dB (2 dB + 4dB coding gain)
$P_0$ (initial probe power)	-60 dBm/Hz (bins 1-50)
	-54 dBm/Hz (bins 51-255)
N (probing steps)	14
$\delta$ (distributed power control margin)	2 dB
Maximum Power	-40 dBm/Hz (bins 1-50)
	-34 dBm/Hz (bins 51-255)
$K_F$ mean	-78.1 dB
$K_F$ standard deviation	11.17 dB
FEXT power sum K	-45.0 dB

The simulation program generates an interference channel model of 48 twisted pairs using the FEXT model in equation (2.5). NEXT is not included in the simulation because the upstream and downstream paths use frequency division duplexing and therefore will not see any NEXT. The frequency response of each twisted pair is a Bellcore standard loop [5] and the coupling is determined by the gamma distributed FEXT interference. Gamma distributed random numbers are generated with an algorithm derived from [35].

As with the wireless system, user arrivals to the network are Poisson distributed and the hold times are exponentially distributed. For each simulation run the load was set to 36 Erlangs, yielding a blocking probability of 1% and an average load of about 36 simultaneous user pairs.

#### **3.4.3** Simulation Results

The simulation program was run for Bellcore standard loops number 1, 7, and 8. These are representative of the worst case conditions which an ADSL system would see in practice. The fixed allocation system is compared with two probing systems, one with a 12 dB SIR margin and the other with a 6 dB SIR margin. The fixed allocation system operates with a 12 dB margin because sometimes the interference level will be worse than the target used for bit allocation. This will not occur with the probing algorithm, however, because power control will maintain every user SIR at the level determined at admission time. Consequently it is possible to reduce the SIR margin without risk of an unacceptable low SIR level due to future users on the same cable.

The results are plotted in Figure 3-6 and summarized in Table 3-4. Even with a 12 dB safety margin, the probing system could out-perform the fixed allocation system in most cases. This demonstrates the effectiveness of probing and adaptive bit allocation for the ADSL wireline system.

Loop	Fixed Allocation Revenue	Probing with 12 dB Margin Revenue	Probing with 6 dB Margin Revenue
1	4608	6091.2	7664.1
7	4608	4609.6	6142.1
8	4608	4590.2	6129.5

 Table 3-4: ADSL Revenue Summary





In general, the power controlled ADSL system with probing allowed each user to increase its data rate by slightly more than one rate class, or roughly 1.5 Mbps. This is equivalent to increasing the average constellation size by 1.5 bits on each bin. For comparison, the gain achieved by channel coding is 6 dB, or 2 bits per symbol per

frequency bin. While significant, this performance gain is not as dramatic as that achieved with the addition of probing and DPCA to the wireless system.

The primary benefit of probing and distributed access for wireline systems is that, like coding, they can provide an increase in bit rate (and revenue) without an increase in system complexity. Power control and bit allocation functions can be encapsulated in the ADSL modem hardware. Since each user makes its own probing and allocation decisions independently, there are no changes required in system software.

## 3.5 Summary

Channel probing can be successfully applied to multiuser systems when it is possible to make very accurate signal and interference measurements. Accurate measurements are possible when three conditions are met: the system symbol rate is rapid compared to the rate at which channel gains vary (either due to fading or mobility), adequate time is allowed for distributed power control to converge completely, and user dynamics are slow enough for probing information to be up to date by the time it is applied. These conditions are met for both high bit rate indoor wireless systems and the DMT ADSL wireline system.

Power control and dynamic channel allocation provide the most benefit in systems that are strongly interference limited with users sharing multiple channels. This fact was demonstrated in [65] and has been verified by the results in this chapter. Wireline systems such as ADSL are limited by maximum constellation sizes and power limits in addition to interference, so they will not benefit as much from dynamic channel allocation and power control as wireless systems. For congested systems, however, the addition of probing, power control, and distributed access can provide higher bit rates and higher revenue for system providers.

## **Chapter 4**

# **Probing Time Varying Channels**

The goal of the previous chapter was to make probing as accurate as possible so individual users could make precise bit allocation decisions. This allowed a system with distributed, dynamic channel allocation to approach the capacity of a centralized system. Unfortunately, this approach will not work when the channel varies faster that it can be estimated. Power control becomes less precise and probing will not be accurate enough to provide bit allocation information. A new algorithm that can produce quick results is required. This chapter explores a faster alternative to precise probing that is more suited to systems with channels that vary. Fast probing is applicable to mobile radio (cellular telephone) and other systems where the radio transceivers are assumed to be moving.

In a power controlled cellular system, the users near cell boundaries transmit the highest power and contribute the greatest amount of interference to users in other cells.

This causes a coupling effect in the network where changes in the interference level in one cell can affect the interference level in other cells that are far away. Coupling can be expressed quantitatively as ||F||, the magnitude of the power control matrix update in equation (3.15). High coupling corresponds to large off-axis terms in F that tend to raise the magnitude [25]. High coupling is undesirable because it will slow the convergence of the power control algorithm and limit its ability to track changes in the channel characteristics. However, since users on the cell boundary experience the largest propagation losses to their base stations, they are forced to operate at or near their maximum power level. Consequently, they do not benefit from power control. These "worst case" users can be moved to fixed assignment channels to reduce the coupling effect. Meanwhile, other users can operate in power controlled channels to yield the capacity advantages of dynamic power and channel allocation. The technique of using two different channel types is called channel partitioning.

Since the channel is varying relatively quickly, it is natural to assume that users will not be able to assign different constellation sizes based on the instantaneous SIR. Rather, they will choose either to use the channel or not, based on whether a fixed minimum SIR level is feasible. The probing problem is then transformed into a problem of finding the best channel to use as quickly as possible. In the worst case, if no power controlled channel can be found, the user can remain in a fixed assignment channel.

The partitioned channel system is especially useful for a cellular mobile system with hand-off. Fixed assignment channels speed the hand-off process, since users do not need to find a power controlled channel immediately when entering a new cell. Also, since new users enter the cell at the boundary, a fixed channel is the appropriate choice. As the user moves away from the boundary, it switches to a power controlled channel to minimize interference.

## 4.1 Channel Partitioning for Mobile Radio

#### **4.1.1 Channel Partitioning Concept**

The cellular system channels are partitioned into two sets, reserved and regular. Reserved channels are assigned to base stations with fixed re-use patterns as described in Section 2.1.2. Regular channels are assigned to every base station and allocated to users by a distributed access algorithm. New users and active users that require large reuse distances occupy reserved channels. All other users occupy regular channels.

There are two operating states for users, probing and active. In the probing state, a user transmits data on a reserved channel while probing a regular channel. In the active state, a user transmits data on a regular channel while applying distributed power control to maintain its SIR. New users enter the network in the probing state and migrate to the active state while transmitting data continuously. Probing is performed in a short segment of each transmission frame so that it will not interfere with data transmission, as illustrated in Figure 4-1. The segmented approach also allows operation with one transmitter and one receiver, minimizing the complexity of the mobile transceiver.

The purpose of channel partitioning is to provide continuous user access while achieving a high level of frequency re-use due to power control. It has higher capacity than fixed re-use systems since most users operate in power controlled channels. The reserved channels allow new users to instantly transmit data instead of waiting for the power control algorithm to find a solution, as is necessary with the algorithms described in Chapter 3. While transmitting on a reserved channel, the new user probes regular channels until it finds one where it can achieve the necessary SIR. If the pool of available reserved channels is full, a new user is blocked. This can be determined instantly, and the user can then make the choice to try again or give up.

Figure 4-1 Signal Frame Format

**Probing State** 

Regular	Reserved
Channel	Channel

Active State

Regular Channel	Regular Channel	
Probing Segment	Transmission Segment	

The channel partitioning system also provides a mechanism for hand-off with power control. When a hand-off becomes necessary, the user moves from a power controlled channel in the old cell to a reserved channel in the new cell. The user then probes for a power controlled channel as if it were a new user. In a similar fashion, a user will hand-off within a cell if a power controlled channel becomes too congested. The ratio of reserved channels to power-controlled channels is selected to maximize the frequency re-use or effective Erlang capacity for a given number of channels. Erlang capacity is maximized when worst case users are forced to remain in the reserved channels while other users occupy power controlled channels. This helps to mitigate the coupling effect and reduces the effect of mobility on network capacity.

Another advantage of the channel partitioning system is backward compatibility with existing fixed channel assignment systems. Since existing cellular radio systems (AMPS and IS-54 TDMA) use fixed assignment, it may be desirable to migrate them to DPCA in the future to increase capacity. Initially, mobile users would use older hardware that would not allow distributed power control. These users would occupy reserved channels only. New mobiles that use both reserved and regular channels would be then be introduced. As users upgrade to newer hardware, reserved channels could be converted to regular channels, increasing system capacity.

The access algorithm is:

- 1. Create a sorted list of accessible base stations with at least one free reserved channel.
- 2. If there are no base stations on the list, block the new user.
- 3. Initiate data transmission on a reserved channel assigned to the nearest base

station.

- 4. Initiate probing on the free regular channel with the lowest interference level.
- 5. If the interference change is greater than the threshold, release the regular channel, hand-off if necessary, and return to step 4.
- 6. Otherwise, wait *N* steps for active users to adapt their power.
- 7. If the SIR level is greater than required, switch to normal state and release the reserved channel.
- 8. Otherwise, release the regular channel and return to step 4.

The power control algorithm is a modification of a distributed algorithm given in Section 2.6.2. The objective is to allow new users to access power controlled channels with less time delay. This reduces the load on reserved channels. The power control update for active users is:

$$p_i(k+1) = \mu_i(k)p_i(k)$$
 (4.1)

where the update factor,  $\mu_i(k)$ , is computed from the ratio of the desired SIR to the measured SIR and upper bounded:

$$\mu_i(k) = \min\left(\frac{\delta\gamma}{r_i(k)}, \mu_{max}\right) \tag{4.2}$$

Here  $\gamma$  is the desired SIR,  $r_i(k)$  is the SIR measured during frame k in the probing segment, and  $\delta$  is the safety-margin. In the steady-state, the SIR of an active user will be maintained at  $\delta\gamma$ . When no users are probing, the interference measured in the probing segment will be entirely due to other active users. In this case, the power
control compensates for changes in signal and interference levels caused by user mobility. The safety margin is determined by the change in the channel gain a user will encounter from frame to frame. If the frame rate is relatively rapid compared to the rate at which the channel changes,  $\delta$  may be small.

Unlike equation (2.17), where new users initially transmit a small power that is increased in fixed steps, the rapid probing algorithm requires users to estimate the power they will require on a power controlled channel. New users attempt access to a power controlled channel by transmitting a fixed power in the probing segment of:

$$p_i(0) = \frac{\delta \delta_{probe} \gamma}{G_{ii} N}$$
(4.3)

where *N* is the receiver noise power,  $G_{ii}$  is the gain to the base station, and  $\delta_{probe}$  is a probing margin factor. The probing margin is used to offset the difference between the received noise power and the eventual interference level if a feasible admission solution is found by the other users. A new user transmits the probing signal for a fixed number of frames and measures the change in interference level in the probing segment. If the change in interference is less than  $\delta$ , it will continue to probe the channel for several steps until the transmitter powers of the active users converge. If the new user achieves the desired SIR, it is admitted to the regular channel. If not, then it tries to probe another channel.

In the distributed power control algorithm with active link protection (described in Section 2.6.2) power updates are always bounded by  $\delta$ . A new users increments its

power in steps of  $\delta$  and active users never need to increase their power by more than  $\delta$  since their SIRs are already greater than or equal to  $\gamma$ . With fixed power probing the situation is different. When a new user probes a channel, it will not directly interfere with active users. However, active users will see increased interference from the power updates of other active users. This occurs because each user updates its power according to the interference seen on the probing segment. If the increase in interference between two frames is greater than the safety margin, an active user's SIR will drop below the desired level. Also, since the transmit power is limited, an active user will sometimes need more power than is possible to maintain its SIR. In either case, if a user cannot maintain its SIR it will switch from the active state to the probing state and probe a new regular channel.

The maximum change in a power update is bounded by  $\mu_{max}$ . The choice of  $\mu_{max} = \delta$  is the most conservative. This will prevent new users from disturbing active users, assuming the active users do not reach their maximum power level. However, it is often better to select a higher level for  $\mu_{max}$ . This allows the active users to adapt to new users more quickly, yielding a smaller probing time. Reduced probing time reduces the load on reserved channels, increasing potential capacity. Active users that fail to maintain their SIRs can re-probe and find new power-controlled channels.

The goal is to maintain a balance between new user access, rapid probing, and active user dropping. In the partitioned channel system, the strategy is to use very rapid

probing for power controlled channel access. This strategy arises from the fact that accurate probing measurement are slow, both because of fading in signals and because of the time required for the power control algorithm to converge. User mobility also limits the accuracy of measurements because channel characteristics will change before they can be measured. Rapid probing places less of a load on the reserved channels than slow probing. The reserved channels then have extra capacity which is used to accommodate active users that are forced out of power controlled channels by imperfect probing. The rapid probing algorithm can allow a user access to a regular channel in 4 to 7 power iterations.

#### 4.1.2 Partitioned System Model

Channel access is evaluated by the load that the system can tolerate at a given blocking probability. Users are assumed to have Poisson arrivals, uniformly distributed over the service area, and exponential hold times. Even under these conditions the blocking probability for the partitioned system is very difficult to compute because the load on the channels cannot be modeled by a simple process with Poisson arrivals. Figure 4-2 presents a schematic model of the system. A new user enters the system and is served by a reserved channel in the nearest base station. The user then moves into a regular channel when it finds one that will allow it to meet the SIR requirement. After time, however, the user may no longer be able to maintain its SIR requirement in the regular channel and it moves back to a reserved channel in the same cell (a re-probe) or a neighboring cell (a hand-off). This occurs because of imperfect probing, finite transmitter dynamic range, and mobility.



Figure 4-2 Network Model

In the system model a new user is blocked when there are no available reserved channels at the closest base station. Preference is given to users that are already in the system by allowing them to access reserved channels in neighboring base stations. An active user will temporarily hand-off to a neighboring cell, if it can achieve the required SIR with that cell's base station. The preference system provides additional reserved channel servers for active users and helps prevent active calls from dropping.

Since reserved channels are used primarily for probing, the hold time will be a multiple of the time required to probe a regular channel. The multiple is equal to the number of attempts it takes to find a suitable regular channel. Users nearer to the base station will probe fewer regular channels and also have the lowest probability of a reprobing event because they will not run out of dynamic range while holding a regular channel. These users will place very little load on the reserved channels. Users near the cell edge, however, will require a longer time to probe for regular channels and tend to re-probe often. These users place the largest load on the reserved channels and will be the primary cause of new user blocking.

The system blocking probability will be a function of the net load on the reserved channels. Roughly speaking, this will be the product of the net arrival rate (new arrivals and re-probes) and the average hold time. If the re-probing rate is low, the input to the reserved channels will be dominated by new arrivals. In this case, the blocking probability can be approximated with the Erlang-B formula [6]. However, as the re-probing rate increases, the arrivals at the reserved channels will no longer be Poisson and this method is not accurate. This occurs because re-probing is a function of the history of the system and the physical location of previous arrivals. Re-probing tends to occur in groups, as regions become congested and the regular channels are no longer able to support their users. Several users will then need to re-probe at the same time. To evaluate the performance under these conditions, it is necessary to use a simulation program.

#### 4.1.3 Trade-offs in System Design

Given the network model, it is possible to examine the trade-offs that determine the user capacity for a given number of channels. In general, the more users that are operating in regular channels, the higher the capacity, since regular channels can accommodate more users. However, enough reserved channels must be set aside to allow new users to probe and active users to re-probe when necessary. A simulation program is used to evaluate the performance of different configurations of channels.

The simulation program evaluates the performance of a network of 19 hexagonal cells, each roughly 7 km in diameter. The radio channel contains an attenuation factor of  $R^{-4}$  and log-normal shadowing with  $\sigma = 8$  dB. Shadowing values are computed on a grid with 100 meter spacing. Actual shadowing values are computed by a 2-dimensional linear interpolation of the shadowing value in decibels.

The total number of channels in the system is 84. Each base station is allocated n reserved channels in a fixed re-use pattern of 7. The remaining 84-7n channels are regular channels that use DPCA. The SIR requirement in regular channels is 18 dB.

New users arrivals are Poisson distributed in time and uniformly distributed in space over the network. Hold times are exponentially distributed, with a mean of 4000 steps. The steps correspond to the power update step in equation (4.1). New users follow the algorithm in Figure 7 for admission, with a waiting time of N=10 steps. Active users in regular channels follow the power update in equation (4.1) with a margin of  $\delta = 1.5$  dB and a maximum power update of  $\mu = 2.0$  dB. Transmitter dynamic range for all users is limited to 66 dB. Users do not move in this first simulation, but they are allowed to temporarily hand-off to a neighboring base station during a re-probe. This gives re-probing active users an access edge over new users and

helps to prevent dropping.

The first simulation run was designed to find the optimal number of reserved channels per cell in an 84 channel system. Five groups of simulations were run with between 1 and 5 reserved channels per cell. For each simulation group, the arrival rate was varied until a blocking probability of 1% was achieved. The results are plotted in Figure 4. The largest user capacity (13.6 Erlangs per cell) was achieved with 2 reserved channels per cell. For the same number of channels, a fixed channel system can provide 12 channels per cell and a user capacity of 5.9 Erlangs per cell. In this case, the partitioned channel system provides a capacity gain of 2.3.

Figure 4-3 User Capacity Optimization



#### 4.2 Mobility and Hand-off

The effects of mobility in fixed re-use systems is analyzed in [19]. In this analysis, the probability of a call failure (blocking during initial access or dropping during hand-off) is shown to be a function of the number of channels, the call load, and the average number of cells visited by a mobile during a call. To maintain a given call failure probability, the number of channels must be increased as either the mobile velocity is increased or the cell size is decreased. The same principle applies to the partitioned channel system. More reserved channels are required to handle hand-offs and probing for new regular channels as the level of mobility increases.

In addition to the problem of hand-off, mobility also introduces complications for the power control and channel access algorithms. As a user moves, its gain terms to the base stations will vary due to propagation distance changes and shadowing effects. As the gains change, new transmit power levels must be computed to maintain the SIR for each user. Eventually, some regular channels will no longer be able to support all of their users and re-probing will occur. This places additional load on the reserved channels and decreases the effectiveness of regular channels to provide capacity gain through power control.

When a user hands-off from one cell to another, there is a chance that a reserved channel will not be available. If this occurs, the user must either be queued (i.e. wait) until a reserved channels is available or forced out of the system (dropped). Dropping users is undesirable, so we choose to queue the active user until a reserved channel is available. The user remains in the old cell on either a reserved channel or a regular channel, if feasible. Hand-off occurs when the new cell has a free reserved channel. The hand-off algorithm is:

- 1. Create a sorted list of accessible base stations.
- 2. If  $G_{MAX} \leq G_C + 3$  dB then do not hand off.
- 3. If the condition in Step 2 is not true for *M* iterations then continue, otherwise cancel the hand off.
- 4. If there is a free reserved channel in the new cell then hand-off.
- 5. Otherwise, wait for 1 power iteration and go back to step 4.

A user starts executing the algorithm when the gain to the current base station,

 $G_C$ , is 3dB less than  $G_{MAX}$ , the maximum gain of all the neighboring base stations. If  $G_{MAX}$  remains at least 3dB greater than  $G_C$  for *M* iterations, the user hands-off to the maximum gain base station. The algorithm provides a hysteresis effect in the hand-off decision to compensate for the effects of shadowing on the cell boundary. Since shadowing makes the cell boundary less distinct, it is desirable to limit unnecessary hand-offs in the boundary region with hysteresis. The threshold value of 3 dB is large enough to avoid the measurement errors that could occur in a practical system. With M = 10 the number of hand-offs in the simulation runs was consistent with the expected number of hand-offs from mobility, indicating proper operation of the algorithm.

The partitioned system assumes a fixed channel re-use factor of K=7 and a

corresponding SIR requirement of 18 dB. This is typical for the AMPS and IS-54 TDMA systems. A total of 84 channels are available with 28 set aside as reserved channels (4 per cell) while the remaining 58 are regular channels. Users in regular channels maintain an SIR of 18 dB. In this simulation, users have 66 dB of dynamic range in their transmitters.

To test the performance of the hand-off algorithm, several sets of simulations have been run. The results are plotted in Figure 4-4.



Figure 4-4 Blocking Probability with Mobility

For comparison, fixed channel results from [19] have been included. These results are for a system with 12 channels per cell. Mobility is characterized by the mobility factor,  $\theta$ , which is the probability that when a user exits a cell it is handing off to another cell.

Both the fixed channel results and the partitioned channel simulation have a mobility factor of  $\theta = 0.5$ . This mobility factor would be considered high for a traditional cellular system, and is designed to demonstrate the performance of the partitioned channel system under stressful conditions. Even under these conditions, the partitioned channel system provides a factor of 2 increase in the user capacity without a reduction in the required SIR level for users.

A pure DPCA system (no fixed re-use channels) would not work well under these high mobility conditions because of the rapid rate at which power controlled channels become infeasible. As channels become infeasible, users must re-probe often. Without a pool of reserved channels, this would require some type of service interruption for every hand-off or re-probe. The length of time to find a new power controlled channel is indeterminate and it is impossible to predict the number and length of the service interruptions that would occur.

# **4.3 Channel Partitioning with Power Level Constraints**

In a practical system, the transmitter power level will be selected from a discrete set:

$$P_k \in \{A_0, A_1, \dots, A_{N-1}\}$$
(4.4)

The levels will usually be in fixed decibel increments. In absolute terms:

$$A_i = \alpha A_{i-1} \tag{4.5}$$

where  $\alpha$  is the increment value. For example, in IS-54 TDMA,  $A_0 = 0$  dBm, N = 8, and  $\alpha = 4$  dB.

With distributed power control, it is necessary to round-up to the next highest discrete power level after the required transmitter power is computed. From equation (4.1), the modified power update is formed:

$$p_i(k+1) = \min A_i \text{ s.t. } A_i \ge \mu_i(k) p_i(k)$$
 (4.6)

A further restriction we can add is to limit the change in power from time k to k + 1 to one power step:

$$\boldsymbol{\mu}_k \in \{ \boldsymbol{\alpha}^{-1}, 1, \boldsymbol{\alpha} \} \tag{4.7}$$

This is useful if the step size is relatively large and more than one step at a time could cause users to be dropped. Another benefit is that only two bits of power control information need to be sent from the base station to the mobile with each power update.

The effect of discrete power levels with distributed power control is to create a distribution of SIR levels between the desired level and the desired level multiplied by the power increment. Under stable conditions the SIR level of a user in a regular channel will be in the range:

$$\delta \gamma \le r_i(k) < \alpha \delta \gamma \tag{4.8}$$

except for users near the base station. These users will have SIRs higher than the upper

bound because they cannot decrease their powers below the minimum power setting,  $A_0$ . The bounds in equation (4.8) arise because feasible power vectors for the users will be quantized to points greater in magnitude than the minimum required. There will be a corresponding loss in the capacity of the regular channels compared to a system with finer power control and greater dynamic range. In the partitioned system, there is a small offset to this capacity loss because the power control will converge faster as the power step size increases. This decreases the load on the reserved channels as new users can probe more quickly. Fewer reserved channels are required and therefore more regular channels are available. These effects are explored below in a simulation of IS-54 TDMA with channel partitioning.

We have investigated the channel partitioning system applied to the uplink (mobile to base-station) of the IS-54 TDMA system. The system we simulated is similar to the one in Section 4.2, however the power control is restricted according to equation (4.5) through equation (4.8), and mobility is decreased somewhat. These changes reflect expected parameters in an upgraded IS-54 system. We examined two possible system upgrades. The first uses probing segments on each user frame. Power updates occur after every other complete TDMA frame, which contains two frames for each user [37]. The second system does not use probing segments but instead uses one out of every ten user frames as a probing frame. Power updates occur after every 5 TDMA frames. In general, the probing segment approach is more desirable since it can provide more accurate power control and faster probing. However, the probing segment approach requires the ability to quickly change frequency, which represents a considerable design change for the mobile unit. The parameters for the simulations are given in Table 4-1.

Cell Diameter	7 Km	
Mobile Velocity	30 mph (ave), 15 mph (s.d.)	
Power Control Update Interval	80 msec (2 frames) with probing segments	
	200 msec (5 frames) without probing seg- ments	
Average Call Time	106 seconds	
Transmitter Dynamic Range	28 dB	
Transmitter Power Steps	8 (4 dB intervals)	
Regular Channels	140	
Fixed Channels	28 (4 per cell)	

 Table 4-1: TDMA Simulation Parameters

In an actual system, there could be up to 1250 channels, based on current frequency allocations for IS-54 TDMA in the United States. We simulated a smaller system (168 channels) to limit the total number of computations. Once again, we simulated a 19 cell network and computed the blocking probability for users in the center cell. Power attenuation was assumed to follow an  $R^{-4}$  characteristic with log-normal shadowing. Shadowing was assumed to be independent every 100 meters and shadowing values were computed by interpolation as in the earlier simulations. When occupying a regular channel, the worst case was 15.5 dB and the average was over 19.5 dB. Fewer than 10% of all users had SIRs lower than 18 dB at any given time. These statistics correspond with the performance of a fixed channel system with K=7 [37]. The results are plotted in Figure 4-5.

Figure 4-5 IS-54 Comparison



A plot of expected fixed assignment performance is included in Figure 4-5 as a comparison. With the operating parameters given in Table 4-1, the mobility parameter is  $\theta = 0.22$ . Based on results in [19] this will result in a capacity reduction of roughly 3%. The fixed assignment curve was computed using the Erlang-B formula [6] with this reduction factor. The partitioned curve was generated from the simulation results of the partitioned system. With call failure probabilities of 1% or less, the probability of a dropped call was less that 0.2%. At 1% failure probability, the partitioned channel system provided a capacity gain of about 60% without probing segments and up to 80% when probing segments are used.

The increase in Erlang capacity of the upgraded IS-54 TDMA system with

channel partitioning (Figure 4-5) was significant, but not as impressive as the capacity gain achieved in the first system simulation (Figure 4-4). This is due to the fact that the limited transmitter dynamic range (28 dB) and quantized power levels (4 dB) of the IS-54 TDMA system cannot provide precise power control. This leads to a wider spread in user SIR levels in the power control channels and a lower average frequency re-use.

### 4.4 Summary

This chapter introduced several new techniques for probing and applying DPCA to systems with time varying channels. In a time varying channel, such as in mobile radio, it is not possible to accurately probe the channel for maximum feasible SIR. Instead, the new user quickly probes to determine whether or not a channel is feasible or not. Separate probing segments (or time slots) are used to protect active users from excessive interference from the new user. The addition of channel partitioning allows users to transmit immediately on a fixed re-use channel while probing for a power controlled channel. This provides instant access for new users and continuous service for active users that must leave power controlled channels due to changes in channel characteristics or hand-offs.

The ratio of fixed channels to power controlled channels is set to maximize the Erlang capacity of the system. The required number of fixed channels depends on the SIR requirement, the amount of mobility, the length of the average connection, and the time required for probing. It is very difficult to determine the optimal channel allocation analytically so simulations are used. For the mobile radio applications discussed here, 1/6 of the available channels were assigned to a fixed re-use pattern.

As with any system that employs DPCA, the frequency re-use and Erlang capacity of the partitioned system depend on the ability to adapt to interference conditions. Consequently, systems that have high dynamic range and fine precision power adjustments can achieve higher capacities. This is consistent with previous DPCA research [39][66].

# Chapter 5

# Probing Channels with Adaptive Antenna Arrays

Adaptive antenna arrays provide three major benefits to a radio communication system. First, they help mitigate the fading effects of a multipath channel by providing antenna diversity. Second, they can help reduce intersymbol interference by combining spatially diverse signals with the desired time delay while rejecting signals with other delays. Third, they help reduce interference from other users by lowering the antenna gain in the direction of interferers. These three benefits are achieved when a minimum mean square error (MMSE) algorithm to adjust the antenna weights on the receiver. Additional benefit is possible with adaptive transmitter antenna arrays.

This chapter examines a new method for probing a channel where each user employs an MMSE adaptive antenna array. As in the previous chapters, it is assumed that distributed power control is employed to allow users to maintain their SIR levels. The combination of adaptive antennas with distributed power control was first investigated in [43]. Here, a formal proof is given of the convergence of power control with the MMSE algorithm. Next, the probing algorithm is derived and its performance is explored.

Probing with adaptive antennas is much more difficult than the probing problem in Chapter 2 because the power control problem is no longer linear and the new user cannot entirely determine the channel SIR characteristics from signal and interference measurements. However, it is possible for the new user to anticipate how well its antenna array can adapt to the desired signal and interferer directions and power levels. This information is then used to estimate the SIR level at a desired transmitter power.

The adaptive antenna probing algorithm is practical because it can quickly predict SIR target levels that are accurate enough for bit allocation algorithms. Simulation results indicate that it works well with varying numbers of antenna elements and interfering users.

# 5.1 Adaptive Antenna Systems for Wireless Communication

Any wireless communication system can be made more efficient through the use of directional antennas. A directional transmitter antenna will increase the amount of desired signal in the direction of the receiver while minimizing the interference to users in other directions. Likewise, a directional receiver antenna will acquire more of the desired signal and reject unwanted signals and noise. For mobile or movable systems it is necessary to point the antenna beam in the desired direction. With an antenna array, it is possible to perform this pointing operation electrically. Each signal from an antenna element is multiplied by a complex weight. The weights, combined with the antenna element patterns, determine the overall antenna pattern [54].

Researchers have explored numerous methods for computing the antenna element weights [16]. Of these, the MMSE method is the most appropriate for a high speed wireless radio modem since it maximizes the signal-to-interference ratio at the receiver. The MMSE solution can be computed with adaptive algorithms, such as leastmean-squares (LMS) and recursive-least-squares (RLS) [29]. Furthermore, MMSE adaptation is compatible with adaptive equalizers that employ the same algorithm.

The simplest multiuser system with antenna arrays is drawn in Figure 5-1.





Each user has a transmitter and receiver array as shown in Figure 5-2 and Figure 5-3.





Figure 5-3 Receiver Array with Weight Update



It is assumed that the transmit and receive antennas are either physically the same or are co-located. Each array has N elements. The receiver computes the MMSE solution for its antenna weights based on measurements of the received signal voltages,

 $x_i$ , on each antenna. In vector form, the antenna voltage can be expressed as the column vector:

$$x = [x_0, x_1, \dots, x_{N-1}]^T$$
(5.1)

Likewise, the receiver weights,  $w^*$ , and the transmitter weights,  $v^*$ , can be expressed the same way. The MMSE solution for the receiver tap weight vector is [29]:

$$w = R^{-1}\rho \tag{5.2}$$

where R is the auto-covariance matrix of the received signal voltage:

$$R = E[xx^H] \tag{5.3}$$

and  $\rho$  is the cross-correlation between the received voltage vector and the desired signal, *d*:

$$\rho = E[xd^*] \tag{5.4}$$

We assume that the MMSE solution is computed with the LMS algorithm because it is the simplest to implement and has robust performance [29]. With the LMS algorithm, the receiver weights are updated on a symbol by symbol basis:

$$w(n+1) = w(n) + \mu x(n)e^{*}(n)$$
(5.5)

where  $\mu$  is scaling factor, and e(n) is the received symbol error (the difference between the desired received symbol and the output of the antenna array):

$$e(n) = d(n) - w^{H}(n)x(n)$$
 (5.6)

The desired symbol, d(n), can be determined either by a training sequence or by a decision device, as shown in Figure 5-3.

The MMSE solution for the receiver weights is employed because it will minimize the interfering signal energy and because it can be computed with well understood algorithms. However, the situation is quite different for computing the transmitter antenna weights. Transmitter weight settings will affect both the received signal and the interference seen by others. Consequently, maximizing the signal power seen at the receiver is optimal for one user but is suboptimal in a multiuser situation. Methods for optimizing the choice of transmitter weights for multiuser systems have been explored in [22][47]. These methods use the received signal covariance matrix, R, to compute transmitter antenna weights that attempt to minimize interference to other users. This is feasible, since by reciprocity the location of received interference is the same as the location where transmitter interference should be minimized.

The simplest of these methods is to use transmitter weights that are normalized versions of the receiver weights [43]:

$$v = \frac{w}{\|w\|} \tag{5.7}$$

While not optimal in all cases, this method is straightforward to implement. The normalization also allows the transmitter weights to be adapted independently of the transmitter power. This help simplify the combination of transmitter antenna weight adaptation and power control.

#### 5.2 Adaptive Antennas with Power Control

If adaptive antennas were perfect at nulling interference, then there would be no

need for power control. Each user would simply null all the interference from other users, yielding a Gaussian channel. The multiuser channel problem would go away completely. In practice, this will never occur for several reasons. First, the number of interferers will often be greater than the number of nulls the antenna array can create. An *N* element antenna array can create only N - 1 nulls. If there are more than N - 1 distinct sources of interference (both from interfering users and multipath), then the MMSE solution will select a compromise that only limits the worst of the interference. Second, the nulls will have finite depth due to finite precision in the weight multipliers and coupling between antenna array elements. Therefore, it will be impossible to completely null any signal there will always be residual interference.

As with the earlier channel model, power control is used to limit the interference from other users and to allow active users to maintain their SIR levels. Once again, it is possible to use individual power control updates for users to compute their power levels in a distributed fashion. First, consider the case when the transmitter antenna weights are fixed. Each user computes an MMSE solution for its receiver weights and updates its transmitter power level,  $p_i(m)$ , according to:

$$p_i(m+1) = \frac{p_i(m)j_{est}(m)}{j_i}$$
(5.8)

where  $j_i$  is the target mean squared error (MSE) level which is related to the SIR target by:

$$j_i = \frac{1}{1 + \gamma_i} \tag{5.9}$$

and  $j_{est}(m)$  is the estimate of the current mean square error. This can be computed by averaging the symbol error e(n). It is important to distinguish the time variables m, the power update time and n the symbol time. A suitable number of symbols (about 100 for 0.5 dB accuracy) are required to average the error between each power level update.

The original distributed power control algorithm given in equation (2.16) computed a matrix inversion by a matrix iteration formula [4]. The conditions for convergence could be determined from the gain matrix, yielding both the maximum feasible SIR for a new user given in equation (3.1) and the residual error due to the finite convergence rate given in equation (3.20). These provided the foundation for precise channel probing. Unfortunately, the power control update with MMSE antenna weights cannot be represented by a matrix iteration, and consequently it is more difficult to derive the conditions for convergence.

However, it is still possible to prove that power control with MMSE antenna adaptation will converge to a feasible solution if a solution exists. The vector of transmitter powers,  $p = [p_0, p_1, ..., p_{M-1}]^T$ , determines the state of the multiuser channel at any point in time. The power updates defined in equation (5.8) form a vector function, I(p), of interference over the power vector. The system has converged when:

$$p = I(p) \tag{5.10}$$

The system will converge if the function I(p) has the properties of positivity, monotonicity, and scalability [67]. Clearly, the MMSE function is always positive. The other properties are less obvious, but also true. A complete proof is given in Appendix A.

## 5.3 Transmitter Antenna Weights

The addition of transmitter weights complicates the power control problem. The state of the system is no longer determined by the vector of transmitter powers, but by a matrix of the transmitter weight vectors of every user sharing the channel. The exact conditions for convergence are unclear, but have been explored via simulation in [43]. In general, the system will converge when the coupling between interfering users is weak. This could be due to the fact that weak coupling implies the system is feasible without transmitter antenna adaptation at all. As the system adapts the receiver weights and power levels towards the fixed point solution, the transmitter antennas adapt to positions that will provide useful antenna gain and allow less transmitter power to be used.

Given the problems in analyzing the transmitter antenna array, it is tempting to leave it out of the probing algorithm. However, the transmitter array is very useful in practice. It supplies antenna gain which means that less transmitter power is required on all communication links. It also reduces interference to other users in the system. For these reasons, it is included in the probing algorithm even if an exact analysis of its convergence properties is incomplete.

## 5.4 Probing with Adaptive Antennas

Probing with adaptive antennas is inherently difficult because it is difficult to know when a feasible solution will exist. The maximum feasible SIR for a new user without adaptive antennas was explored in Section 3.1. This level was determined from the point at which a feasible power control solution was no longer possible. At this point, the intersection of the user SIR constraints given in equation (2.13) moves out to infinity. With adaptive antennas the set of feasible solutions is more complicated to define. As a channel becomes more congested, convergence of the combined adaptive antenna and power control algorithm slows down until it can no longer converge in a useful period of time. A practical method is needed to determine an SIR level that each user can achieve while maintaining timely convergence.

To develop a probing algorithm, the properties of the MSE as seen by a new user are explored. The new user wants to determine the SIR level it can achieve at an access power level, *p*, on a channel with interfering users. It is assumed that a probing channel without any interferers is available for the user to probe. To insure that this channel has the same frequency response characteristics as the desired channel, the probing channel can use a separate time segment or frequency hop on the desired channel. In a frequency hopped system, this could be achieved with a spare hopping pattern.

#### 5.4.1 MSE Bound for a New User

The received signal voltage on the antenna array will be a combination of the desired signal vector,  $x_d$ , the gaussian receiver noise  $x_n$ , and the interference signal  $x_i$ , scaled by a factor  $\sqrt{\beta}$ :

$$x = x_d + x_n + \sqrt{\beta}x_i \tag{5.11}$$

By definition  $\beta = 1$  when the new user is not transmitting on the channel. Once the new user begins transmitting,  $\beta$  will indicate the amount the interference power has increased from this nominal level.

The auto-covariance matrix for the received voltage is:

$$R = E[xx^{H}] = R_{d} + N_{0}I + \beta R_{i}$$
(5.12)

which can be separated into a signal and noise matrix of rank N:

$$R_1 = R_d + N_0 I (5.13)$$

and interference matrix which is approximated to be of rank 1:

$$R_i = bb^H \tag{5.14}$$

This assumes that the interference matrix is dominated by a single user and a single signal path. In an indoor system with light coupling and moderate multipath, this approximation will hold true. The inverse of the total auto-covariance matrix can be computed by a small rank adjustment [30] from the inverse of  $R_1$ :

$$R^{-1} = R_1^{-1} - \frac{R_1^{-1}bb^H R_1^{-1}}{\frac{1}{\beta} + b^H R_1^{-1}b}$$
(5.15)

The cross covariance of the received signal with the desired signal is:

$$a = E[xd^*] \tag{5.16}$$

as in equation (5.4). The MMSE at the receiver will then be:

$$J = 1 - a^H R^{-1} a (5.17)$$

Applying equation (5.15) to equation (5.16) yields:

$$= 1 - a^{H}R_{1}^{-1}a + \frac{\beta a^{H}R_{1}^{-1}bb^{H}R_{1}^{-1}a}{1 + \beta b^{H}R_{1}^{-1}b}$$
(5.18)

The first two terms correspond to the residual MSE without any interference, which will be called  $J_0$ :

$$J_0 = 1 - a^H R_1^{-1} a (5.19)$$

Define

$$q_1 = a^H R_1^{-1} b b^H R_1^{-1} a (5.20)$$

and

$$q_2 = b^H R_1^{-1} b (5.21)$$

Then the MSE as a function of scaled interference power will be:

$$J = J_0 + \frac{\beta q_1}{1 + \beta q_2}$$
(5.22)

and the maximum possible MSE can be upper bounded:

$$J_{max} = \lim_{\beta \to \infty} J = J_0 + \frac{q_1}{q_2}$$
(5.23)

This bound measures the ability of the new user to adapt to changes in the interference level of one resolvable interfering signal. Consequently, it will only apply when a single interferer dominates and the multipath on that signal is relatively low. When these conditions do not apply, the MSE cannot be expressed in the simple form shown in equation (5.22).

#### **5.4.2** Probing with Receiver Measurements

A new user will probe the channel for the values of  $q_1$  and  $q_2$  in order to bound the MSE it can achieve on the channel. To do this, it needs to measure three values,  $J_0$ the MSE on the channel with no interference,  $X_1$  the additional MSE with  $\beta = 1$ , and  $X_2$ , the additional MSE with  $\beta = 2$ . These measurements are made with a modified version of the receiver shown in Figure 5-4.

The new user first measures and stores the interference voltages on each antenna while receiving on the desired channel. It then initiates transmitting on the setup channel (or frequency hop) and adapts the receiving array until convergence. This yields an MSE measurement of  $J_0$ . Next, the stored interference measurements are fed forward into the receive path for each antenna. This simulates the interference present on the desired channel. The array is adapted again until convergence, yielding a second MSE measurement,  $J_1$ .

Figure 5-4 Probing Receiver Structure



The stored interference power is then scaled by a factor of two and the antenna is adapted a third time. The third MSE level is stored in  $J_2$ . Two new quantities are computed:

$$X_1 = J_1 - J_0 \text{ and } X_2 = J_2 - J_0.$$
 (5.24)

From these, the parameters in the MSE bound can be found:

$$_{1} = X_{1} + \frac{2X_{1}^{2} - X_{1}X_{2}}{2(X_{2} - X_{1})}$$
(5.25)

$$q_2 = \frac{2X_1 - X_2}{2(X_2 - X_1)} \tag{5.26}$$

and the ratio has the simplified form:

$$\frac{q_1}{q_2} = \frac{X_2 X_1}{2X_2 - X_1} \tag{5.27}$$

With this information, the new user can predict the MSE it will see on the desired channel as a function of the interference power on that channel.

A simulation plot in Figure 5-5 demonstrates how the probing measurements can be made.



Figure 5-5 Simulation of Probing Algorithm

In this simulation, two pairs of active users, each with a two element antenna array, maintain SIR levels of 13 dB on the channel. A third pair of users probes the channel to determine what SIR level they can achieve. After the active user power levels have stabilized, the new user pair starts probing.

The new user measures  $J_0$ ,  $J_1$ , and  $J_2$  as described above. In this case the SIR

level at  $\beta = 2$  is 9.4 dB while the MSE bound for worst case interference is 6.7 dB. The new user chooses to operate with an SIR of 9.4 dB and adapts to the necessary transmitter power over the course of about 25 frames. The required power level is about 1 dB less than the target level of 40 dB.

This probing technique will work with an arbitrary number of antenna elements and interfering users. It is limited by the fact that one interferer must dominate in order for the MSE bound to hold. However, even when this condition has been met, the bound will tend to under-predict the SIR level that the new user can achieve. In practical systems, the active users will only increase there power levels by a small amount when a new user enters the system. They will often be able to adapt their receivers to null the interference from a new user. They will also adjust their transmitter antenna patterns to help limit interference to the new user. These facts can be incorporated into the probing strategy to make a better algorithm.

#### 5.5 Practical Considerations

A practical probing algorithm must be both fast and precise enough to allow the new user to predict what type of bit allocation it can use on a set of channels. The MSE estimator described in the previous section could not achieve high precision under many conditions because it is impossible to know how much the active users will increase their transmitter powers. Another problem is that accurately measuring the parameters  $X_1$  and  $X_2$  will be difficult because the change in MSE will often be very small.

Accurate measurements take a long time, and this slows down the admissions process.

Given the problems in precisely determining the exact MSE level, it is simpler and faster to make a rough guess of the SIR level that can achieved at the target power level. It turns out that a good estimate is simply:

$$\gamma_{est}(p_{target}) = \frac{1 - J_2}{J_2}$$
(5.28)

where  $p_{target}$  is the target power level and  $J_2$  is the MSE measurement described above.

This estimate works well for lightly coupled systems that employ adaptive transmitter antennas as described in Section 5.3. On un-congested channels where it is possible to achieve a high SIR level, the mean square error will be a loose function of  $\beta$ . (The MSE will be dominated by  $J_0$ .) Consequently, it is not necessary to actually estimate  $\beta$ , but merely to use a value that is reasonable.  $\beta = 2$  is reasonable since in almost all systems with feasible solutions, active users will be able to tolerate the new user without significant increases in their transmitter powers. Except in extreme cases, the increase in interference power will be less than 3 dB.

On more congested channels, the estimate is less accurate. However, in these cases, the estimate tends to under-predict the achievable SIR level. This occurs because active users adapt their transmitter antenna arrays and the actual interference power seen by the new user does not increase as much as predicted.

## 5.6 Simulation Results

A simulation program was developed to test the performance of the estimator in equation (5.28). In the simulation, users probe and are subsequently admitted in pairs. Once admitted, each user follows the power control algorithm to maintain the SIR level it selected during probing. A total of 5 pairs of users are admitted on each run. Each user adapts its receive antenna array with the LMS algorithm and uses normalized receiver weights for transmission as in equation (5.7).

A typical layout for users is shown in Figure 5-6.





Each user pair is randomly located within one of 5 cells in the center of the room. Only

one link is allowed in each cell. This reduces coupling between interfering links which helps to insure convergence of the distributed power control with the adaptive receiver and transmitter antenna arrays. An impulse response is computed between each transmitting antenna element and each receiving element on every user with a raytracing technique described in [43].

The model assumes an operating frequency of 2.45 GHz and a bandwidth of 10 MHz. Antenna arrays vary in size from 2 to 4 elements. The arrays are linear with half-wave spacing. Each array element has an omnidirectional pattern. MSE at each receiver is a combination of both interference from other users and intersymbol interference (ISI).

Each user adapts its receive antenna array with the LMS algorithm on a symbol by symbol basis. MSE is averaged over frame is 256 symbols. After each frame, the transmitter weights are updated according to equation (5.7). Active users also update their transmitter power levels after each frame according to equation (5.8).

The results for a typical simulation run are given in Table 5-1.

User	SIR Estimate	Power Required
0	25 dB	30 dB
1	25	30
2	22	30
3	18	27
4	17	27
5	17	27

 Table 5-1: Typical Simulation Results
User	SIR Estimate	Power Required
6	13	28
7	13	29
8	1	23 (fail)
9	3	30

**Table 5-1: Typical Simulation Results** 

In this simulation, each user has a four element antenna array and a target transmitter power of 30 dB.

A user begins probing by measuring the SIR it can achieve with equation (5.28). The user is then admitted to the channel and increases its power in fixed steps starting at 10 dB until its SIR is at the probing level. It then follows the distributed power control algorithm to maintain its SIR. If the probing is accurate, the new user's power level will be exactly 30 dB after power control is initiated. The third column in Table 5-1 show's the actual power level required to meet the target SIR level. Values over 30 dB are considered probing failures because the user cannot actually achieve the target SIR at the desired power level. Likewise, power values under 27 dB are also considered probing failures because that a higher target SIR (perhaps enough to allocate an additional bit) would be feasible.

A total of 15 sets of simulations were performed with 10 users admitted in each

simulation. The results are given in Table 5-2.

Array Elements	Probing Attempts	Probing Failures
2	50	3
3	50	0
4	50	8

 Table 5-2: Probing Simulation Results,  $p_{target} = 30 \text{ dB}$ 

The probing algorithm was successful 139 times out of 150 admissions, 92% of the time. In general, the algorithm seemed to work best with 3 antennas, but there was no obvious relationship between the number of antennas and the accuracy of the channel probing algorithm.

The simulation program was run a second time to test the performance of the probing algorithm when user interference is more tightly coupled. The results are given in Table 5-2.

Array Elements	Probing Attempts	Probing Failures
2	50	11
3	50	11
4	50	11

Table 5-3: Probing Simulation Results,  $p_{target} = 35 \text{ dB}$ 

The user locations and channel responses were the same as the first run, but the target power level was increased by 5 dB. This allows the first users entering the system to achieve higher SIR levels, at the expense of later users who achieved lower SIR levels

than in the earlier simulation. It also increased the probing failure rate. Out of 150 probes, 117 were successful, or roughly 78%. The probing algorithm was equally accurate for different array sizes.

In general, this simple probing algorithm worked fairly well under moderate interference levels. At higher target power levels, the system coupling became so high that the power control and adaptive antenna algorithm would not always converge. This is consistent with the results in [43] where it was first shown that the adaptive antenna algorithms require light coupling between interferers in order to converge properly. The probing algorithm doesn't place any new restrictions on the use of the adaptive antenna algorithms.

### 5.7 Summary

The goal of channel probing with adaptive antenna arrays is to quickly produce an estimate of the achievable SIR at a target power level. Since the combination of power control and adaptive antennas is very complicated, it is impossible to produce a precise algorithm comparable to the one for power controlled channels. Therefore, the best probing algorithm for most situations is to measure  $J_2$  at the desired transmitter power level as described in Section 5.4.2. The target SIR level can then be computed with equation (5.28). The new user can then begin transmitting on the desired channel at very low power and increase in fixed steps until the target SIR is reached.

In situations where time is less critical, the new user can compute the full set of

parameters described in Section 5.4. Then,  $J_1$  will lower bound the MSE level (upper bound the SIR) and  $J_{max}$  will loosely upper bound the MSE. The new user can begin increasing its transmitter power until it can measure an increase in the interference power from the active users. This interference increase,  $\beta$ , can be substituted into equation (5.22) to estimate the MSE at the target power.

## **Chapter 6**

## Conclusions

The primary purpose of channel probing algorithms is to allow dynamic power and channel allocation to be made distributed. With channel probing, each user will have the information necessary to make its own allocation decisions. If users cooperate, they can use this information to make distributed admission and resource allocation decisions. The exact methods depend on the nature of the communication channel.

Channel probing is most successful when the channel characteristics vary very slowly compared to the symbol rate of the communication system. In this case, the precise probing algorithm described in Chapter 3 can be applied. This algorithm allows the probing user to exactly compute both its maximum feasible SIR level and the SIR level it can achieve as a function of transmitter power. The maximum feasible SIR is the maximum level the user can pursue with distributed power control and still guarantee convergence. The ability to accurately estimate this value means that users will know the exact limits on the bit rate they can achieve on one channel or over a group of channels.

The particular example systems explored in Chapter 3 were a slow frequency hopped CDMA system and a DMT ADSL wireline system. However, the probing technique is also directly applicable to other systems, including direct sequence CDMA. The key feature is that a power control algorithm that equalizes the SIR level must be employed. Since power control algorithms of this type are now well understood it is likely that more systems of this type will be seen in the future. Consequently, more opportunities to use probing will arise.

Chapter 4 explored a wireless system where accurate probing is not possible. This is typical of high mobility systems, such as cellular networks in urban areas. In this case, the access algorithm is modified to determine whether it is appropriate to use a power controlled channel. This type of decision is more robust than precise probing. Even with high mobility, the probing algorithm works well enough to make distributed power control practical.

The addition of adaptive antennas to a power controlled system adds new challenges. These were discussed in Chapter 5. For probing to be feasible, the power controlled system must converge to a fixed point. Otherwise, it is impossible to predict whether or not a feasible solution exists. Convergence to a fixed point was proved for the MMSE adaptive receiver system with power control (Appendix A). When adaptive transmitter antennas are added we know there are situations were convergence is not guaranteed. However, simulation results where normed receiver weights are used for transmission converged consistently when coupling between users was light.

In a system with adaptive transmitter antennas, active users will increase their power when a new user arrives but will often limit their interference in the direction of the new user. This actually helps to simplify the probing process. The new user can assume that the interference will not increase greatly if a channel is feasible. Therefore, it can often estimate the SIR level at a target transmitter power by storing interference measurements and adapting its antenna on a separate setup channel or segment.

There are several directions that look promising for future research work in channel probing. This work examines the cases when very accurate probing is possible and when only simple probing is possible. There are a wide variety of channels that are somewhere between these two extremes. For example, a high bit rate mobile radio channel might benefit from probing for feasible SIR level, but it could not generate the precise results reported in Chapter 2. It could, however, still use probing to help select from a set of channels, rates, and, equalizer settings.

Another area for future work is in adaptive antenna probing. If the receiver employs a more sophisticated antenna processing algorithm than LMS, then a wide variety of signal processing options become available. Sub-space methods could be applied to the received auto-covariance matrix to isolate different interfering signals. This could make probing faster, and allow the receiver to distinguish between ISI and other user interference. Such information would be useful in choosing symbols rates, constellation sizes, and equalizer settings.

As wireless systems become more complex, the need for probing grows. More processing and decision making must be made in a distributed fashion in order to insure both high performance and robustness. However, distributed problem solving requires distributed information. Channel probing allows each user to gather its own information on the state of the channel that it will use. Complicated, information gathering and distributing backbones are simplified or eliminated. Consequently, more resources can be devoted to achieving high capacity.

# **Appendix A**

# MMSE Receiver with Distributed Power Control

### A.1 Background

Yates [67] defines a "standard" distributed power control algorithm as one that will converge to a single fixed point if a feasible solution exists. Below is a proof that minimum mean square error (MMSE) antenna adaptation combined with distributed power control is standard. The constraints for the user transmitter powers are expressed as:

$$p \ge I(p) \tag{A.1}$$

where p is the vector of user transmitter powers and I(p) is the vector function of interference seen by each user:

$$I(p) = (I_1(p), I_2(p), ..., I_N(p))$$
(A.2)

The iterative power control algorithm is:

$$p(t+1) = I(p(t))$$
 (A.3)

and, of course, p > 0. Vector inequalities are defined as strict inequality in all components.

An iteration is standard if the following properties hold:

$$I(p) > 0 \quad \forall p \tag{A.4}$$

if 
$$p \ge p'$$
, then  $I(p) \ge I(p')$  (A.5)

$$\forall \alpha > 1, \ \alpha I(p) > I(\alpha p) \tag{A.6}$$

Now, the interference function for each user is defined to be:

$$I_i(p) = \frac{p_i J_i(p)}{J_i} \tag{A.7}$$

where  $J_i(p)$  is the MMSE at user *i* given power vector *p* and  $J_i$  is the target MMSE for user *i*. At the output of the antenna array the received power is fixed at 1, so the MMSE level can be set by the SIR target level,  $\gamma_i$ :

$$J_i = \frac{1}{\gamma_i + 1} \tag{A.8}$$

The MMSE is represented by:

$$J_i(p) = 1 - r_i^H R^{-1} r_i \tag{A.9}$$

where

$$r_i = E[ud^*] \tag{A.10}$$

is the correlation between the received antenna voltage vector u and the desired signal

d. The signal is normalize to give  $||d||^2 = 1$ . The covariance matrix is defined by:

$$R_i = E[uu^H] \tag{A.11}$$

### A.2 Proof

The update in equation (A.3) using the definition in equation (A.7) will converge to a unique fixed point if a feasible solution exists. This is proved by showing that equation (A.4), equation (A.5), and equation (A.6) are always true.

#### A.1.1 Positivity Property (equation (A.4))

By definition,  $p_i$ ,  $J_i(p)$ , and  $J_i$  are all positive. Consequently, equation (A.7) is positive for all *i*.

#### A.1.2 Monotonicity Property (equation (A.5))

#### A.1.2.1 Case I

For a given user *i* we assume that some  $_{j} > p_{j}'$  for  $j \neq i$ . Then  $I_{i}(p) \ge I_{i}(p')$ because an increase in power from interfering users will always increase the MMSE. When the MMSE increases, the interference function must increase whether or not  $p_{i}$ increases.

#### A.1.2.2 Case II

We will show that  $I_i(p) \ge I_i(p')$  for the case  $j = p_j', j \ne i$  and  $p_i \ge p_i'$ . Start with equation (A.7) and substitute equation (A.9) into the right hand side:

$$I_{i}(p) = \frac{p_{i}}{J_{i}} [1 - r_{i}^{H} R_{i}^{-1} r_{i}]$$
(A.12)

then substitute:

$$r_i = \sqrt{p_i} a_{ii}$$
 and (A.13)

$$_{i} = \sum_{j=1}^{N} p_{j} a_{ij} a_{ij}^{H} + N_{0} I = p_{i} Q_{i} + \sum_{j \neq i} p_{j} Q_{j} + N_{0} I = p_{i} Q_{i} + Q$$
(A.14)

where  $a_{ij}$  are the cross correlation terms from the transmitter antennas to the receiver antenna array and  $Q_j = a_{ij}a_{ij}^H$ . Each  $Q_j$  will have rank equal to the number of resolvable multipath components between the antennas. *I* is the identity matrix. Then:

$$I_{i}(p) = \frac{p_{i}}{J_{i}} [1 - p_{i} a_{ii}^{H} [p_{i} Q_{i} + Q]^{-1} a_{ii}]$$
(A.15)

Applying the matrix inversion lemma to the inner matrix inverse yields:

$$[p_i Q_i + Q]^{-1} = Q^{-1} - \frac{p_i}{1 + p_i q_1} [Q^{-1} Q_i Q^{-1}]$$
(A.16)

where  $q_1 = a_{ii}^H Q^{-1} a_{ii}$ . Note that:

$$a_{ii}^{H}[Q^{-1}Q_{i}Q^{-1}]a_{ii} = a_{ii}^{H}Q^{-1}a_{ii}a_{ii}^{H}Q^{-1}a_{ii} = q_{1}^{2}$$
(A.17)

Applying equation (A.17) to equation (A.15) yields:

$$I_i(p) = \frac{1}{J_i} \left[ p_i - p_i^2 q_1 + \frac{p_i^3 q_1^2}{1 + p_i q_1} \right] = \frac{1}{J_i} \left( \frac{p_i}{1 + p_i q_1} \right)$$
(A.18)

which increases monotonically with  $p_i$ . Therefore,  $I_i(p) \ge I_i(p')$  for the case

$$_j = p_j', j \neq i \text{ and } p_i \geq p_i'.$$

#### A.1.3 Scalability Property (equation (A.6))

Scalability is true if and only if

$$J_i(p) > J_i(\alpha p) \tag{A.19}$$

holds for all *i* with  $\alpha > 1$ . Using the notation from above:

$$J_{i}(p) - J_{i}(\alpha p) = \alpha p_{i} a_{ii}^{H} [Q_{1} - Q_{2}] a_{ii}$$
(A.20)

where

$$Q_1 = \left[Q + \frac{N_0}{\alpha}I\right]^{-1} \text{ and }$$
(A.21)

$$Q_2 = [Q + N_0 I]^{-1}$$
(A.22)

Note that:

$$[Q_1 - Q_2] = N_0 \left(1 + \frac{1}{\alpha}\right) \left[Q^2 + N_0 \left(1 + \frac{1}{\alpha}\right)Q + \frac{N_0^2}{\alpha}I\right]^{-1}$$
(A.23)

The inverse term on the right hand side can be divided into the inverse of the sum of two Hermitian, positive definite components:

$$\left[\left(Q + \frac{N_0}{\alpha}I\right)^2 + \left[N_0\left(1 - \frac{1}{\alpha}\right)Q + N_0\left(\frac{\alpha - 1}{\alpha^2}\right)I\right]\right]^{-1}$$
(A.24)

The first term is square of an auto-covariance matrix and will always be Hermitian, positive definite. The second term is also a valid auto-covariance matrix and will also always be Hermitian, positive definite. Since the sum of two Hermitian, positive definite matrices is Hermitian, positive definite, and the inverse of a Hermitian, positive definite

matrix is also Hermitian, positive definite, the difference  $[Q_1 - Q_2]$  is always positive definite. Therefore equation (A.20) is always positive and the scalability property is proved.

# **Appendix B**

# Glossary

# **B.1** List of Acronyms and Abbreviations

AMPS	Advanced Mobile Phone System
AWGN	Additive white Gaussian noise
BER	Bit error rate
CDMA	Code division multiple access
DCA	Dynamic channel allocation
DPCA	Dynamic power and channel allocation
DPC	Distributed power control
FCC	Federal Communications Commission (U. S.)
FDMA	Frequency division multiple access
FH	Frequency hop

НО	Hand off
Hz	Hertz or cycle per second.
IS-54	Interim Standard 54 (TIA/EIA TDMA cellular standard, U. S.)
IS-95	Interim Standard 95 (TIA/EIA CDMA cellular standard, U. S.)
ISDN	Integrated Services Digital Network
ISI	Inter-symbol interference
ISM	Industrial, Scientific, and Medical (bands, devices)
LAN	Local area network
LOS	Line of sight
MTSO	Mobile telephone switching office
<b>P</b> <sub>block</sub>	Probability of blocking
P <sub>drop</sub>	Probability of dropping
РС	Power control
PCN	Personal Communications Network (Europe)
PCS	Personal Communications Services (U. S.)
QAM	Quadrature amplitude modulation
QOS	Quality of service
RF	Radio frequency
SIR	Signal-to-interference ratio
SNR	Signal-to-noise ratio
SS	Spread Spectrum

ТСМ	Trellis coded modulation
TDMA	Time division multiple access
TIA	Telecommunications Industry Association (U. S.)

## **B.2** Definitions

**Baud**: The transmission rate in symbols per second.

- **Bit error rate:** The ratio of the number of bits incorrectly received to the total number of bits transmitted.
- **Blocking:** New users to the system are declined services due to the lack of channel resources.
- Capacity: Maximum number of users a system can support.
- **Cellular radio**: A system in which a service area is divided into smaller areas called cells where users in each cell communicate with a base station usually located near the center of the cell.
- Channel: A transmission medium for communications.
- **Channel coding:** Adding controlled redundancy to the information sequence to improve reliability of data transmitted through a noisy channel.
- **Coherent detection:** Detection using a reference signal that is synchronized in frequency and phase to the transmitted signal.
- **Code division multiple access:** A way of sharing a common spectrum in which signals from different transmitters are distinguished by a code known to the intended receiver. It is usually divided into two categories: direct sequence (DS) and frequency hopped (FH).
- Dispersion: The spreading, separation, or scatter of a waveform during transmission.
- Diversity: The reception of different versions of the same information bearing signal.
- **Down-link:** The radio link where the base station is transmitting to a user in the coverage area. Also known as the forward link.

- **Dropping:** Users already in the system are denied services due to the lack of channel resources.
- **Dynamic channel allocation (DCA):** A system of allocating channel resources to the user depending on the channel condition and the availability of the channel resource.
- **Dynamic power and channel allocation (DPCA):** A system of assigning channel and power to the user depending on the channel condition and the availability of the channel resource.
- Erlang: A unit-less measure of the offered load to a communications network.
- **Fading:** The variation of the intensity or relative phase of any frequency component of a received signal due to changes in the characteristics of the propagation path with time.
- **Flat fading:** Fading resulting in similar attenuation of all frequency components of signal.
- **Forward link:** The radio link where the base station is transmitting to a user in the coverage area. Also known as the down-link.
- **Frame:** A set of consecutive time slots in which the position of each slot can be identified in reference to the frame start time.
- **Frequency diversity:** A transmission technique that employs different frequency channels to achieve diversity.
- **Frequency hopping (FH):** A spread spectrum technique in which the transmitter rapidly varies its carrier frequency in order to expand the transmission bandwidth.
- **Frequency reuse:** The system of assigning different frequencies to different cells to limit interference while allowing a channel to be used multiple times.
- **Frequency-selective fading:** Fading in which not all frequency components of the received radio signal are attenuated equally.
- **Hand-off (HO):** The process of a user changing the base station it communicates with as it moves across the cell boundaries. Also known as hand-over.
- Interference: Undesired signals in the communication channel.

- **Offered load:** The ratio of the new user arrival rate divided by the system service rate. It may be normalized to the number of channels that are available to the system.
- **Modulation:** The process of varying certain characteristics of a carrier in accordance with an information signal.
- **Multipath:** The large set of propagation paths that the transmitted signal takes to the receiver. The multiple paths could be caused by scattering.
- Multipath fading: Fading that results from multipath.
- **Multiple-access**: A system for users to simultaneously access a common channel resource.
- **Network:** An organization of terminals capable of intercommunication.
- **Outage:** A condition when a user is deprived of service due to unavailability of the communication system.
- **Personal Communication Services (PCS):** For standard purposes, it is an umbrella term to describe services and supporting systems that provide users with the ability to communicate anytime, anywhere, and in any form.
- **Power control (PC):** A technique employed to adjust the transmit power from every radio link to the minimum level required for reliable transmission.
- **Quadrature amplitude modulation:** A coherent digital modulation technique that uses the amplitude in both the I-channel and the Q-channel of the signal to represent information.
- **Receiver:** A device that converts signals used for transmission back to information signals.
- **Reverse link:** The radio link where a user is transmitting to a base station. Also known as the up-link.
- **Signal-to-Interference Ratio (SIR):** The ratio of the desired signal power divided by the total power of the interference.
- **Spread spectrum (SS):** A signaling scheme in which the transmission signal bandwidth is much greater than the information bandwidth.

**Transmitter:** A device that converts information signal to electrical or optical signals

for transmission purposes.

- **Transceiver:** A contraction of "transmitter/receiver." The term is used when a communication device can both transmit and receive.
- **Trellis coded modulation (TCM):** A digital bandwidth-efficient modulation technique that incorporates the concept of set partitioning and channel coding.
- **Up-link:** The radio link where a user is transmitting to a base station. Also known as the reverse link.
- White noise: Noise whose frequency spectrum is uniform over a wide frequency band.
- Wireless communications: General term for communication with electromagnetic waves.

## **Bibliography**

- M. Abramowitz and I. Stegun, Eds., *Handbook of Mathematical Functions*, 9th Ed., Dover, New York: 1970.
- S. V. Ahmed, P. P. Bohn, and N. L. Gottfried, "A tutorial on two-wire digital transmission in the loop plant," *IEEE Trans. on Comm.*, Vol. COM-29, No. 11, pp. 1554-64, Nov. 1981.
- [3] N. Bambos and G. J. Pottie, "Power control based admission policies in cellular radio networks," *Proc. of IEEE Globecomm*, Dec. 6-9, 1992, pp. 863-67, Orlando, FL.
- [4] N. Bambos, S. Chen, and G. J. Pottie, "Radio link admission algorithms for wireless networks with power control," *Proc. of INFOCOM* 1995, April 1995, Boston, MA.
- [5] Bellcore TA-NWT-001210, *Generic Requirements for High Bit Rate Digital Subscriber Lines*, Issue 1, October 1991.
- [6] D. Bertsekas and R. Gallager, *Data Networks, 2nd Ed.*, Englewood, NJ: Prentice-Hall, 1992.
- [7] J. A. C. Bingham, "Multicarrier modulation for data transmission: an idea whose time has come," *IEEE Comm.*, pp. 5-14, May 1990.

- [8] G. Calhoun, *Digital Cellular Radio*, Artech House, Norwood, MA: 1988.
- [9] A. B. Carleial, "Interference channels," *IEEE Trans. on Info. Theory*, Vol. IT-24, No. 1, pp. 60-70, Jan. 1978.
- [10] A. B. Carleial, "Outer bounds on the capacity of interference channels," *IEEE Trans. on Info. Theory*, Vol. IT-29, No. 4, pp. 602-606, July 1983.
- S. C. Chen, N. Bambos, and G. J. Pottie, "On distributed power control for radio networks," *Proc. of IEEE Int. Conf. on Comm.* 1994., pp. 1281-1285, Vol. 3., May 1994.
- [12] R. S. Cheng and S. Verdu, "Gaussian multiaccess channels with ISI: capacity region and multiuser water-filling," *IEEE Trans. on Info. Theory*, Vol. IT-39, No. 3, pp. 773-785, May 1993.
- [13] L. J. Cimini, G. J. Foschini, C.-L. I, "Call blocking performance of distributed algorithms for dynamic channel allocation in microcells," *Proc.* of *IEEE Int. Conf. on Comm.* 1992, pp. 1327-1332.
- [14] L J. Cimini, G. J. Foschini, "Distributed algorithms for dynamic channel allocation in microcellular systems," *Proc. of IEEE Vehicular Technology Conf.* 1992, pp. 641-644, 1992.
- [15] J. M. Cioffi, "A multicarrier primer," Amati Communications Corp. and Stanford University.
- [16] R. T Compton, Jr., *Adaptive Antennas: Concepts and Performance*, Englewood Cliffs, NJ: Prentice-Hall, 1988.
- [17] T. M. Cover and J. A. Thomas, *Elements of Information Theory*, John Wiley

and Sons, Inc., New York, NY, 1991.

- [18] G. J. Foschini and Z. Miljanic. "A Simple distributed autonomous power control algorithm and its convergence," *IEEE Trans. on Vehicular Technology*, Vol. 42, pp. 641-646, 1993.
- [19] G. J. Foschini, B. Gopinath, Z. Milijanic, "Channel cost of mobility," *IEEE Trans. on Vehicular Technology*, Vol. 42, No. 4, pp. 414-24, Nov. 1993.
- [20] G. J. Foschini and Z. Miljanic, "Distributed autonomous wireless channel assignment algorithm with power control," *IEEE Trans. on Vehicular Technology*, Vol. 44, No. 3, pp. 420-29, Aug. 1995.
- [21] F. R. Gantmakher, *The Theory of Matrices*, Chelsea Pub. Co., New York, 1959.
- [22] D. Gerlach and A. Paulraj, "Adaptive transmitting antenna methods for multipath environments," Proc. of Globecom '94, pp. 425-9.
- [23] K. S. Gilhousen, I. M. Jacobs, R. Padovani, A. J. Viterbi, L. A. Weaver, Jr., and C. E. Wheatley, III, "On the capacity of a cellular CDMA system," *IEEE Trans. on Vehicular Technology*, Vol. 40, pp. 303-312, May 1991.
- [24] A. J. Goldsmith, L. J. Greenstein, G. J. Foschini, "Error statistics of realtime power measurements in cellular channels with multipath and shadowing," *IEEE Trans. on Vehicular Technology*, Vol. 43, No. 3, Aug. 1994, pp. 439-446.
- [25] W. Hackbusch, Iterative Solution of Large Sparse Systems of Equations, New York: Springer-Verlag, 1994.

- [26] T. S. Han and K. Kobayashi, "A new achievable rate region for the interference channel," *IEEE Trans. on Info. Theory*, Vol. IT-27, No. 1, pp. 49-60, January 1981.
- [27] C. J. Hansen, C. C. Wang, and G. J. Pottie, "Distributed dynamic channel resource allocation in wireless systems," *Proc. of Asilomar Conference on Signals, Systems, and Computers*, Oct. 1994.
- [28] H. Hashemi, "The indoor radio propagation channel," *Proceedings of the IEEE*, Vol. 81, No. 7, pp. 943-68, July 1993.
- [29] S. Haykin, *Adaptive Filter Theory 2nd Ed.*, Prentice-Hall, Englewood Cliffs, NJ: 1991.
- [30] R. Horn and C. Johnson, *Matrix Analysis*, Cambridge Univ. Press, New York: 1985.
- [31] D. Hughes-Hartogs, United Stated Patent 4,679,227. Jul. 1987.
- [32] W. C. Jakes, Ed., *Microwave Mobile Communications*, IEEE Press, New York: 1974.
- [33] I. Kalet, "The multitone channel," *IEEE Tran. on Comm.*, Vol. 37, No. 2, pp. 119-24, Feb, 1989.
- [34] D. Knuth, *The Art of Computer Programming, Vol. I: Fundamental Algorithms, 2nd Ed.*, Addison-Wesley, Reading, Mass.: 1973.
- [35] D. Knuth, *The Art of Computer Programming, Vol. II: Seminumerical Algorithms, 2nd Ed.,* Addison-Wesley, Reading, Mass.: 1981.
- [36] D. Knuth, The Art of Computer Programming, Vol. III: Sorting and

Searching, Addison-Wesley, Reading, Mass.: 1973.

- [37] W. C. Y. Lee, *Mobile Communications Design Fundamentals*, Wiley and Sons, New York, 1993.
- [38] S. H. Lin, "Statistical behavior for multipair crosstalk," *Bell System Technical Journal*, Vol. 59, No. 6, pp. 955-74, July 1980.
- [39] V. S. Lin and G. J. Pottie, "Implementation of distributed power and admission control for a frequency hopped wireless transceiver," accepted to *IEEE Trans. on Vehicular Technology*.
- [40] D. Mitra, "An asynchronous distributed algorithm for power control in cellular radio systems," Proc. of Fourth Winlab Workshop on Third Generation Wireless Information Network, October 1993.
- [41] A. Papoulis, Probability, Random Variables, and Stochastic Processes, 2nd Ed., McGraw-Hill, New York, 1984.
- [42] D. Parsons, *The Mobile Radio Propagation Channel*, Halstead. New York, 1992.
- [43] E. Perahia, Diversity Combining, Adaptive Antennas, and Equalization for Digital Radio, Ph.D. Thesis, UCLA 1995.
- [44] G. J. Pottie and A. R. Calderbank, "Channel coding strategies for cellular radio," *Proc. of IEEE Int. Symposium on Information Theory*, San Antonio, TX, Jan. 1993. (submitted to *IEEE Trans. on Vehicular Technology*).
- [45] J. G. Proakis, *Digital Communications*, New York: McGraw-Hill, 1983.
- [46] K. Raith, "Capacity of digital cellular TDMA systems," IEEE Trans. on

Vehicular Technology, Vol. 40, No. 2, pp. 323-332, May 1991.

- [47] G. Raleigh, S. Diggavi, V. Jones, and A. Paulraj, "A blind adaptive transmit antenna algorithm for wireless communication," Proc. of ICC '95, vol 3., pp. 1494-9.
- [48] T. S. Rappaport, *Wireless Communications: Principles and Practice*, Prentice-Hall PTR, Upper Saddle River, N.J., 1996.
- [49] E. Senata, *Non-Negative Matrices*. Wiley and Sons: New York, 1973.
- [50] C. E. Shannon, "A mathematical theory of communication," *Bell System Technical Journal*, Vol. 27, No. 3, July 1948.
- [51] M. K. Simon, J. K. Omura, R. A. Scholtz, and B. K. Levitt, *Spread Spectrum Communications Volume I*. Rockville, MD: Computer Science 1985.
- [52] M. K. Simon, J. K. Omura, R. A. Scholtz, and B. K. Levitt, *Spread Spectrum Communications Volume II*. Rockville, MD: Computer Science 1985.
- [53] D. Slepian, "On bandwidth," *Proc. of the IEEE*, Vol. 64, No. 3, March 1976.
- [54] W. L. Stutzman and G. A. Thiele, *Antenna Theory and Design*, New York: John Wiley, 1981.
- [55] T1E1.4/94-007R2, Asymmetric Digital Subscriber Line (ADSL) Working Draft Standard, April 18, 1994.
- [56] G. L. Turin, "Introduction to spread-spectrum antimultipath techniques and their application to urban digital radio," *Proc. of IEEE*, Vol. 68, No. 3, pp. 328-353, Mar. 1980.

- [57] G. L. Turin, "The effects of multipath and fading on the performance of direct-sequence CDMA systems," *IEEE JSAC*, Vol. SAC-2, No. 4, pp. 597-603. July 1984.
- [58] S. Verdu, "Multiuser detection,", *Advances in Statistical Signal Processing*, Vol. 2, pp. 369-409.
- [59] A. J. Viterbi, J. K. Omura, Principles of Digital Communication and Coding. New York: McGraw-Hill, 1979.
- [60] A. J. Viterbi, "When not to spread spectrum a sequel," *IEEE Comm.*, Vol. 23, pp. 12-17, Apr. 1985.
- [61] A. J. Viterbi, A. M. Viterbi, and E. Zehavi, "Performance of powercontrolled wideband terrestrial digital communication," *IEEE Tran. on Comm.*, Vol. 41, No. 4, pp. 559-569, Apr. 1993.
- [62] C. C. Wang, "Dynamic channel resource allocation in frequency hopped wireless communication systems," *Proc. of IEEE Information Theory Symposium*, Norway, 1994.
- [63] C. C. Wang and G. J. Pottie, "Interference avoidance and power control strategies for coded frequency hopped cellular systems," *Proc. of IEEE Int. Conf. on Comm.*, June 1995, pp. 1737-41. Seattle, WA.
- [64] C. C. Wang and G. J. Pottie, "Power control strategies for a coded frequency hopped cellular system," submitted to *IEEE Trans. on Comm.*
- [65] C. C. Wang, *Power Control Strategies and Variable Bit Allocation for FH-CDMA Wireless Systems*, Ph.D. Dissertation, UCLA, 1996.

- [66] J. F. Whitehead, "Performance and capacity of distributed dynamic channel assignment and power control in shadow fading", *Proc. of ICC '93*, May 1993, pp. 910-914.
- [67] R. Yates, "A framework for uplink power control in cellular radio systems," *IEEE Journal on Selected Areas in Communications*, Vol. 13, No. 7, Sept 1995.
- [68] R. D. Yates and C.-Y. Huang, "Integrated power control and base station assignment," *IEEE Trans. on Vehicular Technology*, Vol. 44, pp. 638-44, Aug. 1995.
- [69] J. Zander, "Performance of optimum transmitter power control in cellular radio systems," *IEEE Trans. on Vehicular Technology*, Vol. 41, No. 1, pp. 57-63, Feb. 1992.